UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

		FORM 10-Q		
(Mark One)				
X QUARTERLY REPORT PURSUANT TO SE OF THE SECURITIES EXCHA				
For the quarterly period ended June 30, 2006	5	or		
TRANSITION REPORT PURSUANT TO S SECURITIES EXCHANGE AC	• •	OF THE		
For the transition period from	to			
Commission file number 1-11316				
	OMEG	A HEALTHO	CARE	
	INVI	ESTORS, IN	C.	
	(Exact name of R	egistrant as specified	in its charter)	
	Maryland 38-3	041398		
(Sta	ate of Incorporation)	(I.R.S. Employer Ide	entification No.)	
		ad, Suite 100, Timonion of principal executive		
	(Telephone	(410) 427-1700 number, including ar	ea code)	
Indicate by check mark whether the Act of 1934 during the preceding 12 months to the filing requirements for the past 90 day Yes X No _	(or for such shorter p			
Indicate by check mark whether the "accelerated filer and larger accelerated filer				celerated filer. See definition o
Large accelerated filer	Accelerated file	er <u>X</u>	Non-accelerated filer	
Indicate by check mark whether the	registrant is a shell co o <u>X</u>	ompany (as defined ir	Rule 12b-2 of the Exchange A	Act).
Indicate the number of shares outstanding of	of each of the issuer's	classes of common s	stock as of July 31, 2006.	
Common Stock, \$.10 par value (Class) (Number o	f shares)	58,866,857		

OMEGA HEALTHCARE INVESTORS, INC. FORM 10-Q June 30, 2006

TABLE OF CONTENTS

		Page No.
PART I	Financial Information	
Item 1.	Financial Statements:	
	Consolidated Balance Sheets	2
	Consolidated Statements of Operations (unaudited) Three and six months ended June 30, 2006 and 2005	3
	Consolidated Statements of Cash Flows (unaudited) Six months ended June 30, 2006 and 2005	4
	Notes to Consolidated Financial Statements June 30, 2006 (unaudited)	5
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	17
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	32
Item 4.	Controls and Procedures	33
PART II	Other Information	_
Item 1.	Legal Proceedings	34
Item 1A.	Risk Factors_	34
Item 4.	Submission of Matters to a Vote of Security Holders	34
Item 5.		34
	Other Information	0.7
Item 6.	<u>Exhibits</u>	35

PART I - FINANCIAL INFORMATION

Item 1 - Financial Statements

OMEGA HEALTHCARE INVESTORS, INC. CONSOLIDATED BALANCE SHEETS (in thousands)

	June 30, 2006	December 31, 2005		
	(Unaudited)			
ASSETS	(0,			
Real estate properties				
Land and buildings at cost	\$ 1,060,226	\$	996,127	
Less accumulated depreciation	(171,948)		(157,255)	
Real estate properties - net	888,278		838,872	
Mortgage notes receivable - net	32,381		104,522	
	920,659		943,394	
Other investments - net	29,060		23,490	
	949,719		966,884	
Assets held for sale - net	248		1,243	
Total investments	949,967		968,127	
Cash and cash equivalents	14,053		3,948	
Accounts receivable	6,342		5,885	
Other assets	13,998		37,769	
Total assets	\$ 984,360	\$	1,015,729	
LIABILITIES AND STOCKHOLDERS' EQUITY				
Revolving line of credit	\$ -	\$	58,000	
Unsecured borrowings - net	484,739		505,429	
Other long-term borrowings	41,800		2,800	
Accrued expenses and other liabilities	22,399		19,563	
Operating liabilities for owned properties	125		256	
Total liabilities	549,063		586,048	
Stockholders' equity:				
Preferred stock	118,488		118,488	
Common stock and additional paid-in-capital	680,443		663,607	
Cumulative net earnings	245,843		227,701	
Cumulative dividends paid	(568,150)		(536,041)	
Cumulative dividends - redemption	(43,067)		(43,067)	
Unamortized restricted stock awards	_		(1,167)	
Accumulated other comprehensive income	1,740		160	
Total stockholders' equity	435,297		429,681	
Total liabilities and stockholders' equity	\$ 984,360	\$	1,015,729	

Note - The balance sheet at December 31, 2005 has been derived from the audited consolidated financial statements at that date but does not include all of the information and footnotes required by generally accepted accounting principles for complete financial statements.

See notes to consolidated financial statements.

OMEGA HEALTHCARE INVESTORS, INC. CONSOLIDATED STATEMENTS OF OPERATIONS Unaudited

(in thousands, except per share amounts)

Three Months Ended

Six Months Ended

Revenues 2008 2018 2008			. 00	lung 20				
Revenues 29,042 \$ 22,514 \$ 57,975 \$ 44,262 Mortgage interest income 1,154 1,240 2,338 3,136 Other investment income -net 553 385 1,058 682 Miscollancous 31,061 25,285 61,812 52,452 Total operating revenues 31,061 25,285 61,812 52,452 Expenses 31,061 25,285 61,812 52,452 Expenses 31,061 25,285 61,812 4,235 Ceneral and administrative 2,313 2,123 4,662 4,235 Provision for uncollectible mortgages, notes and accounts's received. 2,313 2,123 4,662 4,235 Loasehold expiration expenses 9,855 9,000 19,722 16,810 Income before other income and expense 21,206 16,285 42,909 35,642 Other income (expense): 11,000 16,285 42,909 35,642 Other income (expense): 11,000 16,943 19,056 13,722			,	-				
Rental income	Revenues	2006	2005	2006	2005			
Mortgage interest income 1,154 1,240 2,338 3,196 Other investment income - net 553 385 1,058 6,431 Miscollanceus 332 1,146 441 4,312 Total operating revenues 31,061 25,285 61,812 52,452 Expenses Bornal operating revenues 31,061 55,285 61,812 52,452 Expenses Bornal operating administrative 2,313 2,123 4,662 4,235 Provision for uncollectible mergages, notes and accounts receivable Bornal operating expenses 83 4,662 4,235 Total operating expenses 9,855 9,000 19,722 156,01 Income before other income and expense 21,266 16,285 42,090 35,642 Other income (expense): 11,000 19,272 1,610 10		\$ 29.042	\$ 22 514	\$ 57 975	\$ 44 262			
Cither Investment Income - net		· · · · · · · · · · · · · · · · · · ·	·		•			
Miscellaneous		-	·	•				
Total operating revenues				·				
Depreciation and amortization 7.542 6.044 15.060 11.742		-						
Depreciation and amortization 7.542 6.044 15.060 11.742	Expenses							
Provision for uncollectible mortgages, notes and accounts receivable 83 750	Depreciation and amortization	7,542	6,044	15,060	11,742			
Case components		2,313	2,123	4,662	4,235			
Total operating expenses	accounts receivable	-	83	-	83			
Income before other income and expense 21,206 16,285 42,090 35,642 Chter income (expense):	Leasehold expiration expense	_	750	_	750			
Components of other comprehensive income	Total operating expenses	9,855	9,000	19,722	16,810			
Interest and other investment income	Income before other income and expense	21,206	16,285	42,090	35,642			
Interest (9,447) (6,948) (19,056) (13,722) Interest - amortization of deferred financing costs (431) (525) (1,074) (1,031) (Other income (expense):							
Interest - amortization of deferred financing costs (431) (525) (1,074) (1,031) Interest - refinancing costs - (3,485) - Provision for impairment on equity securities - (3,360) - (3,360) Total other expense (9,809) (10,809) (23,433) (18,048) Income from continuing operations 11,397 5,476 18,657 17,594 (Loss) from discontinued operations (136) (3,219) (515) (6,033) Net income 11,261 2,257 18,142 11,561 Preferred stock dividends (2,481) (2,864) (4,962) (6,423) Preferred stock conversion and redemption charges - (2,013) - (2,013) Net income (loss) available to common 8,780 (2,620) 13,180 3,125 Income (loss) per common share: Basic:	Interest and other investment income	69	24	182	65			
Interest - refinancing costs		(, ,	(6,948)	, ,	, , ,			
Provision for impairment on equity securities - (3,360) - (3,360) Total other expense (9,809) (10,809) (23,433) (18,048) Income from continuing operations 11,397 5,476 18,657 17,594 (Loss) from discontinued operations (136) (3,219) (515) (6,033) Net income 11,261 2,257 18,142 11,561 Preferred stock dividends (2,481) (2,864) (4,962) (6,423) Preferred stock conversion and redemption charges - (2,013) - (2,013) Net income (loss) available to common \$ 8,780 (2,620) \$ 13,180 \$ 3,125 Income (loss) per common share: S 0.15 0.001 \$ 0.24 0.18 Net income (loss) \$ 0.15 0.005 0.23 0.06 Diluted: S 0.15 0.005 0.24 0.18 Net income (loss) \$ 0.15 0.005 0.24 0.18 Net income (loss) \$ 0.15 0.005 0.024	<u> </u>	(431)	(525)	(1,074)	(1,031)			
Total other expense (9,809) (10,809) (23,433) (18,048)		-	-	(3,485)	-			
Income from continuing operations 11,397 5,476 18,657 17,594	Provision for impairment on equity securities		(3,360)		(3,360)			
Closs from discontinued operations 136 3,219 5,15 6,0033 Net income	Total other expense	(9,809)	(10,809)	(23,433)	(18,048)			
Net income 11,261 2,257 18,142 11,561 Preferred stock dividends (2,481) (2,864) (4,962) (6,423) Preferred stock conversion and redemption charges - (2,013) - (2,013) Net income (loss) available to common \$ 8,780 \$ (2,620) \$ 13,180 \$ 3,125 Income (loss) per common share: Basic: - 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.05 \$ 0.23 \$ 0.06 Diluted: Income from continuing operations \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Dividends declared and paid per common share \$ 0.24 \$ 0.21 \$ 0.47 \$ 0.41 Weighted-average shares outstanding, basic 58,158 </td <td>Income from continuing operations</td> <td>11,397</td> <td>5,476</td> <td>18,657</td> <td>17,594</td>	Income from continuing operations	11,397	5,476	18,657	17,594			
Preferred stock dividends (2,481) (2,864) (4,962) (6,423) Preferred stock conversion and redemption charges - (2,013) - (2,013) Net income (loss) available to common \$ 8,780 \$ (2,620) \$ 13,180 \$ 3,125 Income (loss) per common share: Basic: Income from continuing operations \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.05 \$ 0.23 \$ 0.06 Diluted: Income from continuing operations \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.05 \$ 0.23 \$ 0.06 Diluted: Income from continuing operations \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.05 \$ 0.23 \$ 0.06 Dividends declared and paid per common share \$ 0.15 \$ 0.05 \$ 0.25 \$ 0.23 \$ 0.06 Dividends declared and paid per common share \$ 0.24 \$ 0.21 \$ 0.47 \$ 0.41 Weighted-average shares outstanding, basic \$ 58,158 \$ 51,031 \$ 57,787 \$ 50,980 Weighted-average shares outstanding, diluted \$ 58,237 \$ 51,365 \$ 57,858 \$ 51,339 Components of other comprehensive income: Net income \$ 11,261 \$ 2,257 \$ 18,142 \$ 11,561 Unrealized gain on investments \$ 881 \$ - 1,580 \$ - 1,580	(Loss) from discontinued operations	(136)	(3,219)	(515)	(6,033)			
Preferred stock conversion and redemption charges - (2,013) - (2,013)	Net income	11,261	2,257	18,142	11,561			
Net income (loss) available to common \$ 8,780 \$ (2,013) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) - (2,018) <th< td=""><td>Preferred stock conversion and redemption</td><td>(2,481)</td><td>(2,864)</td><td>(4,962)</td><td>(6,423)</td></th<>	Preferred stock conversion and redemption	(2,481)	(2,864)	(4,962)	(6,423)			
Net income (loss) available to common \$ 8,780 \$ (2,620) \$ 13,180 \$ 3,125 Income (loss) per common share: Basic: Income from continuing operations \$ 0.15 0.01 0.24 0.18 Net income (loss) \$ 0.15 0.05 0.23 0.06 Diluted: Income from continuing operations \$ 0.15 0.01 0.24 0.24 0.18 Net income (loss) \$ 0.15 0.05 0.024 0.24 0.18 Net income (loss) \$ 0.15 0.05 0.024 0.24 0.18 Dividends declared and paid per common share \$ 0.15 0.24 0.05 0.23 0.06 Weighted-average shares outstanding, basic 58,158 51,031 57,787 50,980 Weighted-average shares outstanding, diluted 58,237 51,365 57,858 51,339 Components of other comprehensive income: \$ 11,261 2,257 18,142 11,561 Unrealized gain on investments 881 - 1,580 -	onal goo	_	(2.013)	-	(2.013)			
Basic: Income from continuing operations \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.05 \$ 0.23 \$ 0.06 Diluted: Income from continuing operations \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Dividends declared and paid per common share \$ 0.15 \$ 0.05 \$ 0.23 \$ 0.06 Weighted-average shares outstanding, basic \$ 0.24 \$ 0.21 \$ 0.47 \$ 0.41 Weighted-average shares outstanding, diluted \$ 58,158 \$ 51,031 \$ 57,787 \$ 50,980 Weighted-average shares outstanding, diluted \$ 58,237 \$ 51,365 \$ 57,858 \$ 51,339 Components of other comprehensive income: Net income \$ 11,261 \$ 2,257 \$ 18,142 \$ 11,561 Unrealized gain on investments 881 -< 1,580	Net income (loss) available to common	\$ 8,780						
Income from continuing operations \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18	·							
Net income (loss) \$ 0.15 \$ (0.05) \$ 0.23 \$ 0.06 Diluted: Income from continuing operations \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ (0.05) \$ 0.23 \$ 0.06 Dividends declared and paid per common share \$ 0.24 \$ 0.21 \$ 0.47 \$ 0.41 Weighted-average shares outstanding, basic 58,158 51,031 57,787 50,980 Weighted-average shares outstanding, diluted 58,237 51,365 57,858 51,339 Components of other comprehensive income: \$ 11,261 \$ 2,257 \$ 18,142 \$ 11,561 Unrealized gain on investments 881 - 1,580 -		\$ 0.15	\$ 0.01	\$ 0.24	\$ 0.18			
Diluted: Income from continuing operations \$ 0.15 \$ 0.01 \$ 0.24 \$ 0.18 Net income (loss) \$ 0.15 \$ (0.05) \$ 0.23 \$ 0.06 Dividends declared and paid per common share \$ 0.24 \$ 0.21 \$ 0.47 \$ 0.41 Weighted-average shares outstanding, basic 58,158 51,031 57,787 50,980 Weighted-average shares outstanding, diluted 58,237 51,365 57,858 51,339 Components of other comprehensive income: Net income \$ 11,261 \$ 2,257 \$ 18,142 \$ 11,561 Unrealized gain on investments 881 - 1,580 -	Net income (loss)			\$ 0.23	\$ 0.06			
Net income (loss) \$ 0.15 \$ (0.05) \$ 0.23 \$ 0.06 Dividends declared and paid per common share \$ 0.24 \$ 0.21 \$ 0.47 \$ 0.41 Weighted-average shares outstanding, basic 58,158 51,031 57,787 50,980 Weighted-average shares outstanding, diluted 58,237 51,365 57,858 51,339 Components of other comprehensive income: \$ 11,261 \$ 2,257 \$ 18,142 \$ 11,561 Unrealized gain on investments 881 - 1,580 -								
Net income (loss) \$ 0.15 \$ (0.05) \$ 0.23 \$ 0.06 Dividends declared and paid per common share \$ 0.24 \$ 0.21 \$ 0.47 \$ 0.41 Weighted-average shares outstanding, basic 58,158 51,031 57,787 50,980 Weighted-average shares outstanding, diluted 58,237 51,365 57,858 51,339 Components of other comprehensive income: \$ 11,261 \$ 2,257 \$ 18,142 \$ 11,561 Unrealized gain on investments 881 - 1,580 -	Income from continuing operations	\$ 0.15	\$ 0.01	\$ 0.24	\$ 0.18			
Weighted-average shares outstanding, basic 58,158 51,031 57,787 50,980 Weighted-average shares outstanding, diluted 58,237 51,365 57,858 51,339 Components of other comprehensive income: Net income \$ 11,261 \$ 2,257 \$ 18,142 \$ 11,561 Unrealized gain on investments 881 - 1,580 -	Net income (loss)		\$ (0.05)	\$ 0.23	\$ 0.06			
Weighted-average shares outstanding, diluted 58,237 51,365 57,858 51,339 Components of other comprehensive income: Net income \$ 11,261 \$ 2,257 \$ 18,142 \$ 11,561 Unrealized gain on investments 881 - 1,580 -	Dividends declared and paid per common share	\$ 0.24	\$ 0.21	\$ 0.47	\$ 0.41			
Weighted-average shares outstanding, diluted 58,237 51,365 57,858 51,339 Components of other comprehensive income: Net income \$ 11,261 \$ 2,257 \$ 18,142 \$ 11,561 Unrealized gain on investments 881 - 1,580 -	Mainhtad account about authoridian basis	50.450	54.004	57.707	50.000			
Components of other comprehensive income: Net income \$ 11,261 \$ 2,257 \$ 18,142 \$ 11,561 Unrealized gain on investments 881 - 1,580 -								
Net income \$ 11,261 \$ 2,257 \$ 18,142 \$ 11,561 Unrealized gain on investments 881 - 1,580 -	weighted-average shares outstanding, diluted	58,237	51,365	57,858	51,339			
Unrealized gain on investments 881 - 1,580 -								
			\$ 2,257	•	\$ 11,561			
I otal comprehensive income \$ 12,142 \$ 2,257 \$ 19,722 \$ 11,561	•							
	I otal comprehensive income	<u>\$ 12,142</u>	\$ 2,257	\$ 19,722	<u>\$ 11,561</u>			

OMEGA HEALTHCARE INVESTORS, INC. CONSOLIDATED STATEMENTS OF CASH FLOWS Unaudited (in thousands)

	Six Months Ended
June 30,	June 30,

On any the search state of	2006	2005
Operating activities	Φ 40.44	
Net income	\$ 18,14	2 \$ 11,561
Adjustment to reconcile net income to cash provided by operating activities: Depreciation and amortization (including amounts in discontinued operations)	15,06	9 12,793
Provision for impairment on real estate properties (including amounts in discontinued operations)	12	1 3,700
Provision for uncollectible mortgages, notes and accounts receivable	12	- 83
Provision for impairment on equity securities	_	- 3,360
Refinancing costs	3,48	•
Amortization of deferred financing costs	1,07	
Loss on assets sold - net	38	-
Restricted stock amortization expense	58	*
Other		
	(2:	,
Net change in accounts receivable	(45)	*
Net change in other assets Net change in operating assets and liabilities	1,59	
	2,70	
Net cash provided by operating activities	42,68	34,383
Cash flows from investing activities		
Acquisition of real estate	_	- (120,696)
Proceeds from sale of real estate investments	65	7 24,995
Cash in transit from sale	_	- (12,689)
Capital improvements and funding of other investments	(3,64	9) (1,338)
Proceeds from other investments	17,24	1,262
Investments in other investments	(20,33	9) (5,897)
Collection of mortgage principal	10,39	60,492
Net cash provided by (used in) investing activities	4,30	(53,871)
Cash flows from financing activities		
Proceeds from credit facility borrowings	48,20	168,000
Payments on credit facility borrowings	(106,20	0) (81,500)
Receipts from other long-term borrowings	39,00)
Prepayment of re-financing penalty	(75	5) —
Receipts from dividend reinvestment plan	17,32	757
Receipts/(payments) from exercised options - net	22	
Dividends paid	(32,10	
Redemption of preferred stock	· -	- (50,013)
Payment on common stock offering	(17	
Deferred financing costs paid	(2,38	
Net cash (used in) provided by financing activities	(36,88	
Increase (decrease) in cash and cash equivalents	10,10	5 (11,549)
Cash and cash equivalents at beginning of period	3,94	
Cash and cash equivalents at end of period	\$ 14,05	
·		
Interest paid during the period	\$ 12,64	4 \$ 13,867

OMEGA HEALTHCARE INVESTORS, INC. NOTES TO CONSOLIDATED FINANCIAL STATEMENTS Unaudited June 30, 2006

NOTE 1 - BASIS OF PRESENTATION

The accompanying unaudited consolidated financial statements for Omega Healthcare Investors, Inc. ("Omega" or the "Company") have been prepared in accordance with accounting principles generally accepted in the United States ("GAAP") for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by GAAP for complete financial statements. In our opinion, all adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation have been included. Certain reclassifications have been made to the 2005 financial statements for consistency with the presentation adopted for 2006. Such reclassifications have no effect on previously reported earnings or equity.

In December 2004, the Financial Accounting Standards Board ("FASB") issued FAS No. 123 (revised 2004), *Share-Based Payment* ("FAS No. 123R"), which is a revision of FAS No. 123, *Accounting for Stock-Based Compensation*. FAS No. 123R supersedes Accounting Principles Board ("APB") Opinion No. 25, *Accounting for Stock Issued to Employees*, and amends FAS No. 95, *Statement of Cash Flows*. We adopted FAS No. 123R on January 1, 2006.

In July 2006, the FASB issued FASB Interpretation No. 48, *Accounting for Uncertainty in Income Taxes*, ("FIN 48"). FIN 48 is an interpretation of FASB Statement No. 109, *Accounting for Income Taxes*, and it seeks to reduce the diversity in practice associated with certain aspects of measurement and recognition in accounting for income taxes. In addition, FIN 48 requires expanded disclosure with respect to the uncertainty in income taxes and is effective as of the beginning of our 2007 fiscal year. We are currently evaluating the impact, if any, that FIN 48 will have on our financial statements.

Operating results for the three- and six-month periods ended June 30, 2006 are not necessarily indicative of the results that may be expected for the year ending December 31, 2006. For further information, refer to the financial statements and footnotes included in our annual report on Form 10-K for the year ended December 31, 2005.

Our consolidated financial statements include the accounts of Omega, all direct and indirect wholly owned subsidiaries and one variable interest entity ("VIE") for which we are the primary beneficiary. All inter-company accounts and transactions have been eliminated in consolidation of the financial statements

Financial Accounting Standards Board Interpretation No. 46R, Consolidation of Variable Interest Entities, ("FIN 46R"), addresses the consolidation by business enterprises of VIEs. As a result of the adoption of FIN 46R, we consolidate all VIEs for which we are the primary beneficiary. Generally, a VIE is an entity with one or more of the following characteristics: (a) the total equity investment at risk is not sufficient to permit the entity to finance its activities without additional subordinated financial support; (b) as a group the holders of the equity investment at risk lack (i) the ability to make decisions about an entity's activities through voting or similar rights, (ii) the obligation to absorb the expected losses of the entity, or (iii) the right to receive the expected residual returns of the entity; or (c) the equity investors have voting rights that are not proportional to their economic interests, and substantially all of the entity's activities either involve, or are conducted on behalf of, an investor that has disproportionately few voting rights. FIN 46R requires a VIE to be consolidated in the financial statements of the entity that is determined to be the primary beneficiary of the VIE. The primary beneficiary generally is the entity that will receive a majority of the VIE's expected losses, receive a majority of the VIE's expected residual returns, or both.

In accordance with FIN 46R, we determined that we were the primary beneficiary of one VIE. This VIE is derived from a financing relationship entered into between Omega and one company that is engaged in the ownership and rental of six skilled nursing facilities ("SNFs") and one assisted living facility ("ALF"). The consolidation of the VIE as of June 30, 2006 resulted in an increase in our consolidated total assets (primarily real estate) and liabilities (primarily indebtedness) of approximately \$38.2 million. The creditors of the VIE do not have recourse to our assets.

We have one reportable segment consisting of investments in real estate. Our business is to provide financing and capital to the long-term healthcare industry with a particular focus on skilled nursing facilities located in the United States. Our core portfolio consists of long-term lease and mortgage agreements. All of our leases are "triple-net" leases, which require the tenants to pay all property-related expenses. Our mortgage revenue derives from fixed-rate mortgage loans, which are secured by first mortgage liens on the underlying real estate and personal property of the mortgagor. Substantially all depreciation expenses reflected in the consolidated statements of operations relate to the ownership of our investment in real estate.

NOTE 2 - PROPERTIES

In the ordinary course of our business activities, we periodically evaluate investment opportunities and extend credit to customers. We also regularly engage in lease and loan extensions and modifications. Additionally, we actively monitor and manage our investment portfolio with the objectives of improving credit quality and increasing returns. In connection with portfolio management, we may engage in various collection and foreclosure activities.

If we acquire real estate pursuant to a foreclosure, lease termination or bankruptcy proceeding and do not immediately re-lease or sell the properties to new operators, the assets will be included on the balance sheet as "foreclosed real estate properties," and the value of such assets is reported at the lower of cost or estimated fair value.

The table below summarizes our number of properties and investment by category for the six months ended June 30, 2006:

				Mortgage				Total
		Leased Notes				Facilities		Healthcare
Facility Count	<u> </u>	Property		Receivable		Held for Sale		Facilities
Balance at December 31, 2005		192		32		3		227
Properties sold/mortgages paid		-		(15)		(3)		(18)
Properties transferred to assets held for sale		(1)		-		1		-
Properties transferred to purchase/leaseback		7		(7)		<u>-</u>		<u>-</u>
Balance at June 30, 2006		198	_	10	_	1	_	209
Investment (\$000's)	_							
Balance at December 31, 2005	\$	996,127	\$	104,522	\$	1,243	\$	1,101,892
Properties sold/mortgages paid		-		(48,990)		(1,860)		(50,850)
Properties transferred to assets held for sale		(865)		-		865		-
Properties transferred to purchase/leaseback		61,750		(22,750)		-		39,000
Impairment on properties		(121)		-		-		(121)
Capital expenditures and other		3,335		(401)		<u>-</u>		2,934
Balance at June 30, 2006	\$	1,060,226	\$	32,381	\$	248	\$	1,092,855

Leased Property

Our leased real estate properties, represented by 196 long-term care facilities and two rehabilitation hospitals at June 30, 2006, are leased under provisions of single leases and master leases with initial terms typically ranging from 5 to 15 years, plus renewal options. Substantially all of the leases and master leases provide for minimum annual rentals that are subject to annual increases based upon increases in the

Consumer Price Index ("CPI"). Under the terms of the leases, the lessee is responsible for all maintenance, repairs, taxes and insurance on the leased properties.

Set forth below is a summary of the transactions that occurred in the six months ended June 30, 2006.

Haven Eldercare, LLC

- During the three months ended March 31, 2006, Haven Eldercare, LLC ("Haven"), an existing operator of ours, entered into a \$39 million first mortgage loan with General Electric Capital Corporation ("GE Loan"). Haven used the \$39 million of proceeds to partially repay on a \$62 million mortgage it has with us. Simultaneously, we subordinated the payment of our remaining \$23 million of the mortgage note, due in October 2012, to that of the GE Loan. As a result of this transaction, the interest rate on our remaining mortgage note to Haven rose from 10% to approximately 15%, with annual escalators.
- · In conjunction with the above transactions and the application of FIN 46R, we consolidated the financial statements and related real estate of this Haven entity into our financial statements. The consolidation resulted in the following changes to our consolidated balance sheet as of June 30, 2006: (1) an increase in total gross investments of \$39.0 million; (2) an increase in accumulated depreciation of \$0.8 million; (3) an increase in other long-term borrowings of \$39.0 million; and (4) a reduction of \$0.8 million in cumulative net earnings for the six months ended June 30, 2006 due to the increased depreciation expense. General Electric Capital Corporation and Haven's other creditors do not have recourse to our assets. We have an option to purchase the mortgaged facilities for a fixed price in 2012. Our results of operations reflect the effects of the consolidation of this entity, which is being accounted for similarly to our other purchase-leaseback transactions.

Acquisitions

· There were no acquisitions made during the three and six months ended June 30, 2006.

Assets Sold or Held for Sale

Assets Sold

- · On June 30, 2006, we sold two SNFs in California resulting in an accounting loss of approximately \$0.1 million.
- · On March 31, 2006, we sold a SNF in Illinois resulting in an accounting loss of approximately \$0.2 million.

Held for Sale

- · At June 30, 2006, we had one asset held for sale with a net book value of approximately \$0.2 million.
- During the three months ended March 31, 2006, a \$0.1 million provision for impairment charge was recorded to reduce the carrying value to its sales price of one facility that was under contract to be sold that was subsequently sold during the second quarter of 2006.

Mortgage Notes Receivable

Mortgage notes receivable relate to 10 long-term care facilities. The mortgage notes are secured by first mortgage liens on the borrowers' underlying real estate and personal property. The mortgage notes receivable relate to facilities located in five states, operated by seven independent healthcare operating companies. We monitor compliance with mortgages and when necessary have initiated collection, foreclosure and other proceedings with respect to certain outstanding loans. As of June 30, 2006, we had no foreclosed property, and none of our mortgages were in foreclosure proceedings.

Mortgage interest income is recognized as earned over the terms of the related mortgage notes. Reserves are taken against earned revenues from mortgage interest when collection of amounts due becomes questionable or when negotiations for restructurings of troubled operators lead to lower expectations regarding ultimate collection. When collection is uncertain, mortgage interest income on impaired mortgage loans is recognized as received after taking into account application of security deposits.

Hickory Creek Healthcare Foundation, Inc.

On June 16, 2006, we received approximately \$10 million in proceeds on a mortgage loan payoff. We held mortgages on 15 facilities located in Indiana, representing 619 beds.

Haven Eldercare, LLC.

During the three months ended March 31, 2006, Haven used the \$39 million of proceeds from the GE Loan to partially repay on a \$62 million mortgage it has with us. Simultaneously, we subordinated the payment of its remaining \$23 million on the mortgage note to that of the GE Loan (see Note - 2 Properties; Leased Property).

NOTE 3 - CONCENTRATION OF RISK

As of June 30, 2006, our portfolio of domestic investments consisted of 209 healthcare facilities, located in 27 states and operated by 34 third-party operators. Our gross investment in these facilities, net of impairments and before reserve for uncollectible loans, totaled approximately \$1.1 billion at June 30, 2006, with approximately 98% of our real estate investments related to long-term care facilities. This portfolio is made up of 196 long-term healthcare facilities, two rehabilitation hospitals owned and leased to third parties, fixed rate mortgages on 10 long-term healthcare facilities and one facility held for sale. At June 30, 2006, we also held miscellaneous investments of approximately \$29 million, consisting primarily of secured loans to third-party operators of our facilities.

At June 30, 2006, approximately 25% of our real estate investments were operated by two public companies: Sun Healthcare Group, Inc. ("Sun") (15%) and Advocat Inc., ("Advocat") (10%). Our largest private company operators (by investment) were CommuniCare Health Services, Inc. ("CommuniCare") (18%), Haven (11%), Guardian LTC Management, Inc. (7%) and Essex Healthcare Corporation (7%). No other operator represents more than 5% of our investments. The three states in which we had our highest concentration of investments were Ohio (25%), Florida (10%) and Pennsylvania (9%) at June 30, 2006.

For the three-month period ended June 30, 2006, our revenues from operations totaled \$31.1 million, of which approximately \$5.8 million were from Sun (19%), \$4.6 million from CommuniCare (15%) and \$3.2 million from Advocat (10%). For the six-month period ended June 30, 2006, our revenues from operations totaled \$61.8 million, of which approximately \$11.6 million were from Sun (19%), \$9.2 million from CommuniCare (15%) and \$6.3 million from Advocat (10%). No other operator generated more than 10% of our revenues from operations for the three- and six-month periods ended June 30, 2006.

Sun and Advocate are subject to the reporting requirements of the Securities and Exchange Commission (the "Commission") and are required to file with the Commission annual reports containing audited financial information and quarterly reports containing unaudited interim financial information. Sun's and Advocat's filings with the Commission can be found at the Commission's website at www.sec.gov. We are providing this data for information purposes only, and you are encouraged to obtain Sun's and Advocat's publicly available filings from the Commission.

NOTE 4 - DIVIDENDS

In order to qualify as a REIT, we are required to distribute dividends (other than capital gain dividends) to our stockholders in an amount at least equal to (A) the sum of (i) 90% of our "REIT taxable income" (computed without regard to the dividends paid deduction and our net capital gain), and (ii) 90% of the net income (after tax), if any, from foreclosure property, minus (B) the sum of certain items of non-cash income. In addition, if we dispose of any built-in gain asset during a recognition period, we will be required to distribute at least 90% of the built-in gain (after tax), if any, recognized on the disposition of such asset. Such distributions must be paid in the taxable year to which they relate, or in the following taxable year if declared before we timely file our tax return for such year and paid on or before the first regular dividend payment after such declaration. In addition, such distributions are required to be made pro rata, with no preference to any share of stock as compared with other shares of the same class, and with no preference to one class of stock as compared with another class except to the extent that such class is entitled to such a preference. To the extent that we do not distribute all of our net capital gain or do distribute at least 90%, but less than 100% of our "REIT taxable income," as adjusted, we will be subject to tax thereon at regular ordinary and capital gain corporate tax rates. In addition, our \$200 million revolving senior secured credit facility ("Credit Facility") has certain financial covenants that limit the distribution of dividends paid during a fiscal quarter to no more than 95% of our aggregate cumulative funds from operations ("FFO") as defined in the loan agreement governing the Credit Facility (the "Loan Agreement"), unless a greater distribution is required to maintain REIT status. The Loan Agreement defines FFO as net income (or loss) plus depreciation and amortization and shall be adjusted for charges related to: (i) restructuring our debt; (ii)

Common Dividends

On July 17, 2006, the Board of Directors announced a common stock dividend of \$0.24 per share to be paid August 15, 2006 to common stockholders of record on July 31, 2006.

On April 18, 2006, the Board of Directors declared a common stock dividend of \$0.24 per share, an increase of \$0.01 per common share compared to the prior quarter. The common dividend was paid May 15, 2006 to common stockholders of record on April 28, 2006.

On January 17, 2006, the Board of Directors declared a common stock dividend of \$0.23 per share, an increase of \$0.01 per common share compared to the prior quarter. The common stock dividend was paid February 15, 2006 to common stockholders of record on January 31, 2006.

Series D Preferred Dividends

On July 17, 2006, the Board of Directors declared the regular quarterly dividends for the 8.375% Series D Cumulative Redeemable Preferred Stock ("Series D Preferred Stock") to stockholders of record on July 31, 2006. The stockholders of record of the Series D Preferred Stock on July 31, 2006 will be paid dividends in the amount of \$0.52344 per preferred share on August 15, 2006. The liquidation

preference for our Series D Preferred Stock is \$25.00 per share. Regular quarterly preferred dividends for the Series D Preferred Stock represent dividends for the period May 1, 2006 through July 31, 2006.

On April 18, 2006, the Board of Directors declared regular quarterly dividends of approximately \$0.52344 per preferred share on the Series D Preferred Stock that were paid May 15, 2006 to preferred stockholders of record on April 28, 2006.

On January 17, 2006, the Board of Directors declared regular quarterly dividends of approximately \$0.52344 per preferred share on the Series D Preferred Stock that were paid February 15, 2006 to preferred stockholders of record on January 31, 2006.

NOTE 5 - TAXES

As a qualified REIT, as long as we distribute 100% of our taxable income, we will not be subject to Federal income taxes on our income, and no provisions for Federal income taxes have been made. We are permitted to own up to 100% of a "taxable REIT subsidiary" ("TRS"). Currently, we have two TRSs that are taxable as corporations and that pay federal, state and local income tax on their net income at the applicable corporate rates. These TRSs had net operating loss carry-forwards as of June 30, 2006 of \$10.1 million. These loss carry-forwards were fully reserved with a valuation allowance due to uncertainties regarding realization.

NOTE 6 - STOCK-BASED COMPENSATION

Stock Options

Prior to January 1, 2006, we accounted for stock based compensation using the intrinsic value method as defined by APB Opinion No. 25, *Accounting for Stock Issued to Employees*. Effective January 1, 2006, we adopted FAS No. 123R using the modified prospective method. Accordingly, we have not restated prior period amounts. The additional expense to be recorded in 2006 as a result of this adoption is approximately \$3 thousand. Under the provisions of FAS No. 123R, the "Unamortized restricted stock awards" line on our consolidated balance sheet, a contra-equity line representing the amount of unrecognized share-based compensation costs, is no longer presented. Accordingly, for the six-month period ended June 30, 2006, the amount that had been on the "Unamortized restricted stock awards" line was reversed through the "Common stock and additional paid-in-capital" line on our consolidated balance sheet.

Under the terms of our 2000 Stock Incentive Plan (the "2000 Plan"), we reserved 3,500,000 shares of common stock. The exercise price per share of an option under the 2000 Plan cannot be reduced after the date of grant, nor can an option be cancelled in exchange for an option with a lower exercise price per share. The 2000 Plan provides for non-employee directors to receive options that vest over three years while other grants vest over the period required in the agreement applicable to the individual recipient. Directors, officers, employees and consultants are eligible to participate in the 2000 Plan. At June 30, 2006, there were outstanding options for 52,581 shares of common stock granted to eight eligible participants under the 2000 Plan. Additionally, 355,655 shares of restricted stock have been granted under the provisions of the 2000 Plan, and as of June 30, 2006, there were no shares of unvested restricted stock outstanding under the 2000 Plan.

At June 30, 2006, under the 2000 Plan, there were outstanding options for 50,912 shares of common stock granted to eight participants currently exercisable with a weighted-average exercise price of \$13.58, with exercise prices ranging from \$2.96 to \$37.20. There were 559,960 shares available for future grants as of June 30, 2006. A breakdown of the options outstanding under the 2000 Plan as of June 30, 2006, by price range, is presented below:

Option Price			ghted erage	Weighted Average Remaining Life	Number	Weighted verage Price on Options
Range	Number	Exerci	se Price	(Years)	Exercisable	 Exercisable
\$2.96 -\$3.81	11,918	\$	3.41	5.51	11,918	\$ 3.41
\$6.02 -\$9.33	22,330	\$	6.67	6.06	20,661	\$ 6.46
\$20.25 -\$37.20	18,333	\$	28.23	2.02	18,333	\$ 28.23

On April 20, 2004, our Board of Directors approved the 2004 Stock Incentive Plan (the "2004 Plan"), which was subsequently approved by our stockholders at our annual meeting held on June 3, 2004. Under the terms of the 2004 Plan, we reserved 3,000,000 shares of common stock. The exercise price per share of an option under the 2004 Plan cannot be less than fair market value (as defined in the 2004 Plan) on the date of grant. The exercise price per share of an option under the 2004 Plan cannot be reduced after the date of grant, nor can an option be cancelled in exchange for an option with a lower exercise price per share. Directors, officers, employees and consultants are eligible to participate in the 2004 Plan. As of June 30, 2006, a total of 348,695 shares of restricted stock and 317,500 restricted stock units have been granted under the 2004 Plan, and as of June 30, 2006, there were no outstanding options to purchase shares of common stock under the 2004 Plan.

At June 30, 2006, the only options outstanding to purchase shares of our common stock were options issued under our 2000 Plan for 52,581 shares of common stock. For the quarter ended June 30, 2006, no options were granted under any of our stock incentive plans. The following is a summary of option activity under the 2000 Plan:

Stock Options	Number of Shares	<u>E</u> :	xercise Price	eighted- rage Price	Weighted- Average Remaining Contractual Term
Outstanding at December 31, 2005	227,440	\$\$	2.760-37.205	\$ 5.457	4.6
Granted during 1st quarter 2006					
Exercised	(174,191)		2.760-9.330	2.979	_
Cancelled	(668)		22.452-22.452	 22.452	
Outstanding at March 31, 2006	52,581	\$\$	2.960-37.205	\$ 13.448	4.4
Granted during 2 nd quarter 2006				 	
Exercised	_			_	_
Cancelled				 	
Outstanding at June 30, 2006	52,581	\$\$	2.960-37.205	\$ 13.448	4.2
Vested at June 30, 2006	50,912	\$\$	2.960-37.205	\$ 13.583	4.0

Non-Vested Options	Number of Shares	Ex	ercise Price	eighted- age Price	Average Remaining Contractual Term
Non-vested at December 31, 2005	74,985	\$\$	2.760-9.330	\$ 3.200	7.0
Vested during 1 st quarter 2006	(73,316)		2.760-9.330	 3.059	
Non-vested at March 31, 2006	1,669	\$\$	9.330-9.330	\$ 9.330	7.8
Vested during 2 nd quarter 2006				_	
Non-vested at June 30, 2006	1,669	\$\$	9.330-9.330	\$ 9.330	7.5

Weighted-

Cash received from exercise under all stock-based payment arrangements for the six months ended June 30, 2006 and 2005 was \$0.9 million and \$0.1 million, respectively. Cash used to settle equity instruments granted under stock-based payment arrangements for the six months ended June 30, 2006 and 2005 was \$0.7 million and \$0.9 million, respectively.

In 2005, we accounted for our stock-based compensation arrangements in accordance with the intrinsic value method as defined by Accounting Principles Board Opinion (APB) No. 25, "Accounting for Stock Issued to Employees." The following table presents the effect on net income and earnings per share if we had applied the fair value recognition provisions of FAS No. 123R to our stock-based compensation.

The reported and pro forma net income and earnings per share figures for 2006 in the table are the same because share-based compensation expense is calculated under the provisions of FAS No. 123R. The 2006 amounts are included in the table below to provide detail for comparative purposes to the 2005 amounts.

	Three Months Ended June 30,			Six Months Ended June 30,				
		2006		2005		2006		2005
			(in th	ousands, except	per	share amounts)		
Net income (loss) to common stockholders Add: Stock-based compensation expense included	\$	8,780	\$	(2,620)	\$	13,180	\$	3,125
in net income (loss) to common stockholders		292		285		585		571
		9,072		(2,335)		13,765		3,696
Less: Stock-based compensation expense determined under the fair value based method								
for all awards		292		342		585		690
Pro forma net income (loss) to common stockholders	\$	8,780	\$	(2,677)	\$	13,180	\$	3,006
Earnings (loss) per share:								
Basic, as reported	\$	0.15	\$	(0.05)	\$	0.23	\$	0.06
Basic, pro forma	\$	0.15	\$	(0.05)	\$	0.23	\$	0.06
Diluted, as reported	\$	0.15	\$	(0.05)	\$	0.23	\$	0.06
Diluted, pro forma	\$	0.15	\$	(0.05)	\$	0.23	\$	0.06

Restricted Stock

On September 10, 2004, we entered into restricted stock agreements with four executive officers under the 2004 Plan. A total of 317,500 shares of restricted stock were granted, which equated to approximately \$3.3 million of deferred compensation. The shares vest thirty-three and one-third percent

(33 1/3%) on each of January 1, 2005, January 1, 2006 and January 1, 2007 so long as the executive officer remains employed on the vesting date, with vesting accelerating upon a qualifying termination of employment or upon the occurrence of a change of control (as defined in the applicable restricted stock agreements). As a result of the grant, we recorded \$0.3 million and \$0.6 million of non-cash compensation expense for the three and six-month periods ended June 30, 2006 and 2005, respectively.

For the six-month period ended June 30, 2006, we issued 1,822 shares of restricted common stock to each non-employee director and an additional 2,000 shares of restricted common stock to the Chairman of the Board under the 2004 Plan for a total of 11,110 shares. These shares represent a payment of the portion of the directors' annual retainer that is payable in shares of our common stock.

As of June 30, 2006, there was \$670 thousand of total unrecognized compensation cost related to these restricted stock awards.

Performance Restricted Stock Units

On September 10, 2004, we entered into performance restricted stock unit agreements with our four executive officers under the 2004 Plan. A total of 317,500 restricted stock units were issued under the 2004 Plan and will fully vest into shares of common stock when our company attains \$0.30 per share of adjusted funds from operations (as defined in the applicable restricted stock unit agreements) for two (2) consecutive quarters, with vesting accelerating upon a qualifying termination of employment or upon the occurrence of a change of control (as defined in the applicable restricted stock unit agreements). The performance restricted stock units expire on December 31, 2007 if the performance criteria has not been met. The issuance of restricted stock units has no impact on our calculation of diluted earnings per common share at this time; however, under our current method of accounting for stock-based compensation, the expense related to the restricted stock units will be recognized when it becomes probable that the vesting requirements will be met.

NOTE 7 - FINANCING ACTIVITIES AND BORROWING ARRANGEMENTS

Bank Credit Agreements

At June 30, 2006, we had no outstanding borrowings under our \$200 million revolving senior secured credit facility (the "New Credit Facility"); however, \$2.9 million was utilized for the issuance of letters of credit, leaving availability of \$197.1 million. The New Credit Facility, entered into on March 31, 2006, is being provided by Bank of America, N.A., as Administrative Agent, Deutsche Bank Trust Company Americas, UBS Securities LLC, General Electric Capital Corporation, LaSalle Bank N.A., and Citicorp North America, Inc. and will be used for acquisitions and general corporate purposes.

The New Credit Facility replaced our previous \$200 million senior secured credit facility (the "Prior Credit Facility"), that was terminated on March 31, 2006. We will realize a 125 basis point savings on LIBOR-based loans under the New Credit Facility, as compared to LIBOR-based loans under our Prior Credit Facility. The New Credit Facility matures on March 31, 2010, and includes an "accordion feature" that permits us to expand our borrowing capacity to \$300 million during our first two years.

For the three-month period ending March 31, 2006, we recorded a one-time, non-cash charge of approximately \$2.7 million relating to the write-off of deferred financing costs associated with the termination of our Prior Credit Facility.

Our long-term borrowings require us to meet certain property level financial covenants and corporate financial covenants, including prescribed leverage, fixed charge coverage, minimum net worth, limitations on additional indebtedness and limitations on dividend payouts. As of June 30, 2006, we were in compliance with all property level and corporate financial covenants.

\$100 Million Aggregate Principal Amount of 6.95% Unsecured Notes Tender and Redemption

On December 16, 2005, we initiated a tender offer and consent solicitation for all of our outstanding \$100 million aggregate principal amount 6.95% notes due 2007 (the "2007 Notes"). On December 30, 2005, we accepted for purchase 79.3% of the aggregate principal amount of the 2007 Notes outstanding that were tendered. On December 30, 2005, our Board of Directors also authorized the redemption of all outstanding 2007 Notes that were not otherwise tendered. On December 30, 2005, upon our irrevocable funding of the full redemption price for the 2007 Notes and certain other acts required by the Indenture governing the 2007 Notes, the Trustee of the 2007 Notes certified in writing to us (the "Certificate of Satisfaction and Discharge") that the Indenture was satisfied and discharged as of December 30, 2005, except for certain administrative provisions. In accordance with FASB Statement No. 140, *Accounting for Transfers and Servicing of Financial Assets and Extinguishment of Liabilities*, ("FAS 140") we removed 79.3% of the aggregate principal amount of the 2007 Notes, which were tendered in our tender offer and consent solicitation, and the corresponding portion of the funds held in trust by the Trustee to pay the tender price from our balance sheet and recognized \$2.8 million of additional interest expense associated with the tender offer. On January 18, 2006, we completed the redemption of the remaining 2007 Notes not otherwise tendered. Accordingly, we reduced other assets, representing the funds deposited with the Trustee, and unsecured borrowings by \$21 million. In connection with the redemption and in accordance with FAS 140, we recognized \$0.8 million of additional interest expense in the first quarter of 2006. As of January 18, 2006, none of the 2007 Notes remained outstanding.

Other Long-Term Borrowings

During the three months ended March 31, 2006, Haven used the \$39 million of proceeds from the GE Loan to partially repay a portion of a \$62 million mortgage it has with us. Simultaneously, we subordinated the payment of its remaining \$23 million on the mortgage note to that of the GE Loan. In conjunction with the above transactions and the application of FIN 46R, we consolidated the financial statements of this Haven entity into our financial statements, which contained the long-term borrowings with General Electric Capital Corporation of \$39.0 million. The loan has an interest rate of approximately seven percent and is due in 2012. The lender of the \$39.0 million does not have recourse to our assets (see Note - 2 Properties; Leased Property).

NOTE 8 - LITIGATION

We are subject to various legal proceedings, claims and other actions arising out of the normal course of business. While any legal proceeding or claim has an element of uncertainty, management believes that the outcome of each lawsuit, claim or legal proceeding that is pending or threatened, or all of them combined, will not have a material adverse effect on our consolidated financial position or results of operations.

We and several of our wholly-owned subsidiaries have been named as defendants in professional liability claims related to our former owned and operated facilities. Other third-party managers responsible for the day-to-day operations of these facilities have also been named as defendants in these claims. In these suits, patients of certain previously owned and operated facilities have alleged significant damages, including punitive damages against the defendants. The majority of these lawsuits representing the most significant amount of exposure were settled in 2004. There currently is one lawsuit pending that is in the discovery stage, and we are unable to predict the likely outcome of this lawsuit at this time.

NOTE 9 - DISCONTINUED OPERATIONS

Statement of Financial Accounting Standards ("SFAS") No. 144, *Accounting for the Impairment or Disposal of Long-Lived Assets,* requires the presentation of the net operating results of facilities sold during 2006 or currently classified as held-for-sale as income from discontinued operations for all periods presented. We incurred a net loss from discontinued operations of approximately \$0.1 million and \$0.5 million for the three- and six-month periods ended June 30, 2006, respectively, in the accompanying consolidated statements of operations.

The following table summarizes the results of operations of facilities sold or held-for-sale during the three and six months ended June 30, 2006 and 2005, respectively.

	Three Months Ended June 30,			Six Months Ended			
				Ju			
	2	006		2005	2006		2005
				(in thou	sands)		
Revenues							
Rental income	\$	_	\$	1,430	\$ —	\$	2,896
Other income		_		12			24
Subtotal revenues				1,442		·	2,920
Expenses							
Depreciation and amortization		_		496	9	١	1,051
General and administrative		3		_	4		_
Provision for impairment					121		3,700
Subtotal expenses	-	3		496	134	_	4,751
Income (loss) before loss on sale of assets		(3)		946	(134	.)	(1,831)
Loss on assets sold - net		(133)		(4,165)	(381)	(4,202)
Loss from discontinued operations	\$	(136)	\$	(3,219)	\$ (515) \$	(6,033)

NOTE 10 - EARNINGS PER SHARE

The computation of basic earnings per common share ("EPS") is computed by dividing net income available to common stockholders by the weighted-average number of shares of common stock outstanding during the relevant period. Diluted EPS reflects the potential dilution that could occur from shares issuable through stock-based compensation, including stock options and restricted stock rewards.

For the three- and six-month periods ended June 30, 2006 and 2005, the dilutive effect from stock options was immaterial.

NOTE 11 - SUBSEQUENT EVENTS

On August 1, 2006, we completed a transaction with Litchfield Investment Company, LLC and its affiliates ("Litchfield") to purchase 30 skilled nursing facilities and one independent living center for a total investment of approximately \$171 million. The facilities total 3,847 beds and are located in the states of Colorado (5), Florida (7), Idaho (1), Louisiana (13), and Texas (5). The facilities were subject to master leases with three national healthcare providers, which are existing tenants of the Company. The tenants are Home Quality Management, Inc. ("HQM"), Nexion Health, Inc. ("Nexion"), and Peak Medical Corporation, which was acquired by Sun in December of 2005. We used a combination of cash on hand and \$150 million of New Credit Facility borrowings to finance the Litchfield transaction.

Simultaneously with the close of the purchase transaction, the seven HQM facilities were combined into an Amended and Restated Master Lease containing 13 facilities between us and HQM. In addition, the 18 Nexion facilities were combined into an Amended and Restated Master Lease containing 22 facilities between us and Nexion.

We entered into a Master Lease, Assignment and Assumption Agreement with Litchfield on the six Sun facilities. These six facilities are currently under a master lease that expires on September 30, 2007.

The total incremental annualized rent from the 31 facility transaction will be approximately \$17.1 million.

Item 2 - Management's Discussion and Analysis of Financial Condition and Results of Operations

Forward-looking Statements, Reimbursement Issues and Other Factors Affecting Future Results

The following discussion should be read in conjunction with the financial statements and notes thereto appearing elsewhere in this document. This document contains forward-looking statements within the meaning of the federal securities laws, including statements regarding potential financings and potential future changes in reimbursement. These statements relate to our expectations, beliefs, intentions, plans, objectives, goals, strategies, future events, performance and underlying assumptions and other statements other than statements of historical facts. In some cases, you can identify forward-looking statements by the use of forward-looking terminology including, but not limited to, terms such as "may," "will," "anticipates," "expects," "believes," "intends," "should" or comparable terms or the negative thereof. These statements are based on information available on the date of this filing and only speak as to the date hereof and no obligation to update such forward-looking statements should be assumed. Our actual results may differ materially from those reflected in the forward-looking statements contained herein as a result of a variety of factors, including, among other things:

- (i) those items discussed under "Risk Factors" in Item 1A to our annual report on Form 10-K for the year ended December 31, 2005;
- (ii) uncertainties relating to the business operations of the operators of our assets, including those relating to reimbursement by third-party payors, regulatory matters and occupancy levels;
- (iii) the ability of any operators in bankruptcy to reject unexpired lease obligations, modify the terms of our mortgages and impede our ability to collect unpaid rent or interest during the process of a bankruptcy proceeding and retain security deposits for the debtors' obligations;
- (iv) our ability to sell closed assets on a timely basis and on terms that allow us to realize the carrying value of these assets;
- (v) our ability to negotiate appropriate modifications to the terms of our credit facility;
- (vi) our ability to manage, re-lease or sell any owned and operated facilities;
- (vii) the availability and cost of capital;
- (viii) competition in the financing of healthcare facilities;
- (ix) regulatory and other changes in the healthcare sector;
- (x) the effect of economic and market conditions generally and, particularly, in the healthcare industry;
- (xi) changes in interest rates;
- (xii) the amount and yield of any additional investments;
- (xiii) changes in tax laws and regulations affecting real estate investment trusts; and
- (xiv) changes in the ratings of our debt and preferred securities.

Overview

At June 30, 2006, our portfolio of domestic investments consisted of 209 healthcare facilities, located in 27 states and operated by 34 third-party operators. Our gross investment in these facilities, net of impairments and before reserve for uncollectible loans, totaled approximately \$1.1 billion at June 30, 2006, with approximately 98% of our real estate investments related to long-term care facilities. This portfolio is made up of 196 long-term healthcare facilities, two rehabilitation hospitals owned and leased to third parties, fixed rate mortgages on 10 long-term healthcare facilities and one long-term healthcare facility that is currently held for sale. At June 30, 2006, we also held other investments of approximately \$29 million, consisting primarily of secured loans to third-party operators of our facilities.

Medicare Reimbursement

All of our properties are used as healthcare facilities; therefore, we are directly affected by the risk associated with the healthcare industry. Our lessees and mortgagors, as well as any facilities that may be owned and operated for our own account from time to time, derive a substantial portion of their net operating revenues from third-party payors, including the Medicare and Medicaid programs. These programs are highly regulated by federal, state and local laws, rules and regulations and are subject to frequent and substantial change.

In 1997, the Balanced Budget Act significantly reduced spending levels for the Medicare and Medicaid programs, in part because the legislation modified the payment methodology for skilled nursing facilities ("SNFs") by shifting payments for services provided to Medicare beneficiaries from a reasonable cost basis to a prospective payment system. Under the prospective payment system, SNFs are paid on a per diem prospective case-mix adjusted basis for all covered services. Implementation of the prospective payment system has affected each long-term care facility to a different degree, depending upon the amount of revenue such facility derives from Medicare patients.

Legislation adopted in 1999 and 2000 provided for a few temporary increases to Medicare payment rates, but these temporary increases have since expired. Specifically, in 1999 the Balanced Budget Refinement Act included a 4% across-the-board increase of the adjusted federal per diem payment rates for all patient acuity categories (known as "Resource Utilization Groups" or "RUGs") that were in effect from April 2000 through September 30, 2002. In 2000, the Benefits Improvement and Protection Act included a 16.7% increase in the nursing component of the case-mix adjusted federal periodic payment rate, which was implemented in April 2000 and also expired October 1, 2002. The October 1, 2002 expiration of these temporary increases has had an adverse impact on the revenues of the operators of SNFs and has negatively impacted some operators' ability to satisfy their monthly lease or debt payments to us.

The Balanced Budget Refinement Act and the Benefits Improvement and Protection Act also established temporary increases, beginning in April 2001, to Medicare payment rates to SNFs that were designated to remain in place until the Centers for Medicare and Medicaid Services ("CMS") implemented refinements to the existing RUG case-mix classification system to more accurately estimate the cost of non-therapy ancillary services. The Balanced Budget Refinement Act provided for a 20% increase for 15 RUG categories until CMS modified the RUG case-mix classification system. The Benefits Improvement and Protection Act modified this payment increase by reducing the 20% increase for three of the 15 RUGs to a 6.7% increase and instituting an additional 6.7% increase for eleven other RUGs.

On August 4, 2005, CMS published a final rule, effective October 1, 2005, establishing Medicare payments for SNFs under the prospective payment system for federal fiscal year 2006 (October 1, 2005 to September 30, 2006). The final rule modified the RUG case-mix classification system and added nine new categories to the system, expanding the number of RUGs from 44 to 53. The implementation of the RUG refinements triggered the expiration of the temporary payment increases of 20% and 6.7% established by the Balanced Budget Refinement Act and the Benefits Improvement and Protection Act, respectively. Additionally, CMS announced updates in the final rule to reimbursement rates for SNFs in federal fiscal year 2006 based on an increase in the "full market-basket" of 3.1%.

In the August 4, 2005 notice, CMS estimated that the increases in Medicare reimbursements to SNFs arising from the refinements to the prospective payment system and the market basket update under the final rule will offset the reductions stemming from the elimination of the temporary increases during federal fiscal year 2006. CMS estimated that there will be an overall increase in Medicare payments to SNFs totaling \$20 million in fiscal year 2006 compared to 2005.

On July 27, 2006, CMS posted a notice updating the payment rates to SNFs for fiscal year 2007 (October 1, 2006 to September 30, 2007). The market basket increase factor is 3.1% for 2007. CMS

estimates that the payment update will increase aggregate payments to SNFs nationwide by approximately \$560 million in fiscal year 2007 compared to 2006.

Nonetheless, we cannot accurately predict what effect, if any, these changes will have on our lessees and mortgagors in 2006 and beyond. These changes to the Medicare prospective payment system for SNFs, including the elimination of temporary increases, could adversely impact the revenues of the operators of nursing facilities and could negatively impact the ability of some of our lessees and mortgagors to satisfy their monthly lease or debt payments to us.

A 128% temporary increase in the per diem amount paid to SNFs for residents who have AIDS took effect on October 1, 2004. This temporary payment increase arises from the Medicare Prescription Drug Improvement and Modernization Act of 2003 ("Medicare Modernization Act"). The August 2005 notice announcing the final rule for the SNF prospective payment system for fiscal year 2006 clarified that the increase will remain in effect for fiscal year 2006, although CMS also noted that the AIDS add-on was not intended to be permanent.

A significant change enacted under the Medicare Modernization Act is the creation of a new prescription drug benefit, Medicare Part D, which went into effect January 1, 2006. The significant expansion of benefits for Medicare beneficiaries arising under the expanded prescription drug benefit could result in financial pressures on the Medicare program that might result in future legislative and regulatory changes with impacts for our operators. As part of this new program, the prescription drug benefits for patients who are dually eligible for both Medicare and Medicaid are being transitioned from Medicaid to Medicare, and many of these patients reside in long-term care facilities. The Medicare program has experienced significant operational difficulties in transitioning prescription drug coverage for this population since the benefit went into effect on January 1, 2006, although it is unclear whether or how issues involving Medicare Part D might have any direct financial impacts on our operators.

On February 8, 2006, the President signed into law a \$39.7 billion budget reconciliation package called the Deficit Reduction Act of 2005 ("Deficit Reduction Act") to lower the federal budget deficit. The Deficit Reduction Act includes estimated net savings of \$8.3 billion from the Medicare program over 5 years.

The Deficit Reduction Act contained a provision reducing payments to SNFs for allowable bad debts. Previously, Medicare reimbursed SNFs for 100% of beneficiary bad debt arising from unpaid deductibles and coinsurance amounts. In 2003, CMS released a proposed rule seeking to reduce bad debt reimbursement rates for certain providers, including SNFs, by 30% over a three-year period. CMS never finalized its 2003 proposal. The Deficit Reduction Act reduced payments to SNFs for allowable bad debts by 30% effective October 1, 2005 for those individuals not dually eligible for Medicare and Medicaid. Bad debt payments for the dually eligible population will remain at 100%. These reductions in Medicare payments for bad debt could have a material adverse effect on our operators' financial condition and operations, which could adversely affect their ability to meet their payment obligations to us.

The Deficit Reduction Act also contained a provision governing the therapy caps that went into place under Medicare on January 1, 2006. The therapy caps limit the physical therapy, speech-language therapy and occupation therapy services that a Medicare beneficiary can receive during a calendar year. The therapy caps were in effect for calendar year 1999 and then suspended by Congress for three years. An inflation-adjusted therapy limit (\$1,590 per year) was implemented in September of 2002, but then once again suspended in December of 2003 by the Medicare Modernization Act. Under the Medicare Modernization Act, Congress placed a two-year moratorium on implementation of the caps, which expired at the end of 2005.

The inflation-adjusted therapy caps are set at \$1,740 for 2006. These caps do not apply to therapy services covered under Medicare Part A in a SNF, although the caps apply in most other instances involving patients in SNFs or long-term care facilities who receive therapy services covered under

Medicare Part B. The Deficit Reduction Act permitted exceptions in 2006 for therapy services to exceed the caps when the therapy services are deemed medically necessary by the Medicare program. The implementation of the therapy caps could have a material adverse effect on our operators' financial condition and operations, which could adversely affect their ability to meet their payment obligations to us.

In general, we cannot be assured that federal reimbursement will remain at levels comparable to present levels or that such reimbursement will be sufficient for our lessees or mortgagors to cover all operating and fixed costs necessary to care for Medicare and Medicaid patients. We also cannot be assured that there will be any future legislation to increase Medicare payment rates for SNFs, and if such payment rates for SNFs are not increased in the future, some of our lessees and mortgagors may have difficulty meeting their payment obligations to us.

Medicaid and Other Third-Party Reimbursement

Each state has its own Medicaid program that is funded jointly by the state and federal government. Federal law governs how each state manages its Medicaid program, but there is wide latitude for states to customize Medicaid programs to fit the needs and resources of their citizens. Currently, Medicaid is the single largest source of financing for long-term care in the United States. Rising Medicaid costs and decreasing state revenues caused by recent economic conditions have prompted an increasing number of states to cut or consider reductions in Medicaid funding as a means of balancing their respective state budgets. Existing and future initiatives affecting Medicaid reimbursement may reduce utilization of (and reimbursement for) services offered by the operators of our properties.

In recent years, many states have announced actual or potential budget shortfalls, and many budget forecasts in 2006 could be similar. As a result of these budget shortfalls, many states have announced that they are implementing or considering implementing "freezes" or cuts in Medicaid reimbursement rates, including rates paid to SNF and long-term care providers, or reductions in Medicaid enrollee benefits, including long-term care benefits. We cannot predict the extent to which Medicaid rate freezes, cuts or benefit reductions ultimately will be adopted, the number of states that will adopt them or the impact of such adoption on our operators. However, extensive Medicaid rate cuts, freezes or benefit reductions could have a material adverse effect on our operators' liquidity, financial condition and operations, which could adversely affect their ability to make lease or mortgage payments to us.

The Deficit Reduction Act included \$4.7 billion in estimated savings from Medicaid and the State Children's Health Insurance Program over five years. The Deficit Reduction Act gives states the option to increase Medicaid cost-sharing and reduce Medicaid benefits, accounting for an estimated \$3.2 billion in federal savings over five years. The remainder of the Medicaid savings under the Deficit Reduction Act comes primarily from changes to prescription drug reimbursement (\$3.9 billion in savings over five years) and tightened policies governing asset transfers (\$2.4 billion in savings over five years).

Asset transfer policies, which determine Medicaid eligibility based on whether a Medicaid applicant has transferred assets for less than fair value, are more restrictive under the Deficit Reduction Act, which extends the look-back period to five years, moves the start of the penalty period and makes individuals with more than \$500,000 in home equity ineligible for nursing home benefits (previously, the home was excluded as a countable asset for purposes of Medicaid eligibility). These changes could have a material adverse effect on our operators' financial condition and operations, which could adversely affect their ability to meet their payment obligations to us.

Additional reductions in federal funding are expected for some state Medicaid programs as a result of changes in the percentage rates used for determining federal assistance on a state-by-state basis. Legislation has been introduced in Congress that would partially mitigate the reductions for some states that would experience significant reductions in federal funding, although whether Congress will enact this or other legislation remains uncertain.

Finally, private payors, including managed care payors, increasingly are demanding discounted fee structures and the assumption by healthcare providers of all or a portion of the financial risk of operating a healthcare facility. Efforts to impose greater discounts and more stringent cost controls are expected to continue. Any changes in reimbursement policies that reduce reimbursement levels could adversely affect the revenues of our lessees and mortgagors, thereby adversely affecting those lessees' and mortgagors' abilities to make their monthly lease or debt payments to us.

Fraud and Abuse Laws and Regulations

There are various extremely complex and largely uninterpreted federal and state laws governing a wide array of referrals, relationships and arrangements and prohibiting fraud by healthcare providers, including criminal provisions that prohibit filing false claims or making false statements to receive payment or certification under Medicare and Medicaid, and failing to refund overpayments or improper payments. The federal and state governments are devoting increasing attention and resources to anti-fraud initiatives against healthcare providers. Penalties for healthcare fraud have been increased and expanded over recent years, including broader provisions for the exclusion of providers from the Medicare and Medicaid programs. The Office of the Inspector General for the U.S. Department of Health and Human Services, in cooperation with other federal and state agencies, continues to focus on the activities of SNFs in certain states in which we have properties.

In addition, the federal False Claims Act allows a private individual with knowledge of fraud to bring a claim on behalf of the federal government and earn a percentage of the federal government's recovery. Because of these monetary incentives, these so-called "whistleblower" suits have become more frequent. Some states currently have statutes that are analogous to the federal False Claims Act. The Deficit Reduction Act encourages additional states to enact such legislation and may encourage increased enforcement activity by permitting states to retain 10% of any recovery for that state's Medicaid program if the enacted legislation is at least as rigorous as the federal False Claims Act. The violation of any of these laws or regulations by an operator may result in the imposition of fines or other penalties that could jeopardize that operator's ability to make lease or mortgage payments to us or to continue operating its facility.

Legislative and Regulatory Developments

Each year, legislative and regulatory proposals are introduced or proposed in Congress and state legislatures as well as by federal and state agencies that, if implemented, could result in major changes in the healthcare system, either nationally or at the state level. In addition, regulatory proposals and rules are released on an ongoing basis that may have major impacts on the healthcare system generally and the industries in which our operators do business. Legislative and regulatory developments can be expected to occur on an ongoing basis at the local, state and federal levels that have direct or indirect impacts on the policies governing the reimbursement levels paid to our facilities by public and private third-party payors, the costs of doing business and the threshold requirements that must be met for facilities to continue operation or to expand.

The Medicare Modernization Act, which is one example of such legislation, was enacted in December 2003. The significant expansion of other benefits for Medicare beneficiaries under this Act, such as the prescription drug benefit, could create financial pressures on the Medicare program that might result in future legislative and regulatory changes with impacts on our operators. Although the creation of a prescription drug benefit for Medicare beneficiaries was expected to generate fiscal relief for state Medicaid programs, the structure of the benefit and costs associated with its implementation may mitigate the relief for states that originally was anticipated.

The Deficit Reduction Act is another example of such legislation. The provisions in the legislation designed to create cost savings from both Medicare and Medicaid could diminish reimbursement for our operators under both Medicare and Medicaid.

CMS also launched the Nursing Home Quality Initiative program in 2002, which requires nursing homes participating in Medicare to provide consumers with comparative information about the quality of care at the facility. CMS plans to initiate a new Nursing Home Quality campaign this fall to be conducted over the next two years with the ultimate goal being improvement in quality of life and efficiency of care delivery. In the event any of our operators do not maintain the same or superior levels of quality care as their competitors, patients could choose alternate facilities, which could adversely impact our operators' revenues. In addition, the reporting of such information could lead in the future to reimbursement policies that reward or penalize facilities on the basis of the reported quality of care parameters.

In late 2005, CMS began soliciting public comments regarding a demonstration to examine pay-for-performance approaches in the nursing home setting that would offer financial incentives for facilities to deliver high quality care. In June 2006, Abt Associates published recommendations for CMS on how to design this demonstration project. The proposed three-year demonstration could begin as early as late 2006 or early 2007, with CMS inviting state agencies to participate in the project. Other proposals under consideration include efforts by individual states to control costs by decreasing state Medicaid reimbursements in the current or future fiscal years and federal legislation addressing various issues, such as improving quality of care and reducing medical errors throughout the health care industry. We cannot accurately predict whether specific proposals will be adopted or, if adopted, what effect, if any, these proposals would have on operators and, thus, our business.

Critical Accounting Policies and Estimates

Our financial statements are prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP") and a summary of our significant accounting policies is included in Note 2 to our annual report on Form 10-K for the year ended December 31, 2005. Our preparation of the financial statements requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities, disclosure of contingent assets and liabilities at the date of our financial statements, and the reported amounts of revenue and expenses during the reporting period.

We have identified five significant accounting policies that we believe are critical accounting policies. These critical accounting policies are those that have the most impact on the reporting of our financial condition and those requiring significant assumptions, judgments and estimates. With respect to these critical accounting policies, we believe the application of judgments and assessments is consistently applied and produces financial information that fairly presents the results of operations for all periods presented. The five critical accounting policies are:

Revenue Recognition

With the exception of certain master leases, rental income and mortgage interest income are recognized as earned over the terms of the related master leases and mortgage notes, respectively. Such income generally includes periodic increases based on pre-determined formulas (i.e., such as increases in the CPI) as defined in the master leases and mortgage loan agreements. Reserves are taken against earned revenues from leases and mortgages when collection becomes questionable or when negotiations for restructurings of troubled operators result in significant uncertainty regarding ultimate collection. The amount of the reserve is estimated based on what management believes will likely be collected. When collection is uncertain, lease revenues are recorded when received, after taking into account application of security deposits. Interest income on impaired mortgage loans is recognized when received after taking into account application of principal repayments and security deposits.

We recognize the minimum base rental revenue under master leases with fixed increases on a straight-line basis over the term of the related lease. Accrued straight-line rents represent the rental revenue recognized in excess of rents due under the lease agreements at the balance sheet date.

Gains on sales of real estate assets are recognized pursuant to the provisions of SFAS No. 66, *Accounting for Sales of Real Estate*. The specific timing of the recognition of the sale and the related gain is measured against the various criteria in SFAS No. 66 related to the terms of the transactions and any continuing involvement associated with the assets sold. To the extent the sales criteria are not met, we defer gain recognition until the sales criteria are met.

Depreciation and Asset Impairment

Under GAAP, real estate assets are stated at the lower of depreciated cost or fair value, if deemed impaired. Depreciation is computed on a straight-line basis over the estimated useful lives of 20 to 40 years for buildings and improvements and three to ten years for furniture, fixtures and equipment. Management periodically, but not less than annually, evaluates our real estate investments for impairment indicators, including the evaluation of our assets' useful lives. The judgment regarding the existence of impairment indicators is based on factors such as, but not limited to, market conditions, operator performance and legal structure. If indicators of impairment are present, management evaluates the carrying value of the related real estate investments in relation to the future undiscounted cash flows of the underlying facilities. Provisions for impairment losses related to long-lived assets are recognized when expected future undiscounted cash flows are determined to be permanently less than the carrying values of the assets. An adjustment is made to the net carrying value of the leased properties and other long-lived assets for the excess of historical cost over fair value. The fair value of the real estate investment is determined by market research, which includes valuing the property as a nursing home as well as other alternative uses. All impairments are taken as a period cost at that time, and depreciation is adjusted going forward to reflect the new value assigned to the asset.

If we decide to sell rental properties or land holdings, we evaluate the recoverability of the carrying amounts of the assets. If the evaluation indicates that the carrying value is not recoverable from estimated net sales proceeds, the property is written down to estimated fair value less costs to sell. Our estimates of cash flows and fair values of the properties are based on current market conditions and consider matters such as rental rates and occupancies for comparable properties, recent sales data for comparable properties, and, where applicable, contracts or the results of negotiations with purchasers or prospective purchasers.

During the six months ended June 30, 2006, we recognized an impairment loss associated with one facility of approximately \$0.1 million.

Loan Impairment

Management, periodically but not less than annually, evaluates our outstanding loans and notes receivable. When management identifies potential loan impairment indicators, such as non-payment under the loan documents, impairment of the underlying collateral, financial difficulty of the operator or other circumstances that may impair full execution of the loan documents, and management believes these indicators are permanent, then the loan is written down to the present value of the expected future cash flows. In cases where expected future cash flows cannot be estimated, the loan is written down to the fair value of the collateral. The fair value of the loan is determined by market research, which includes valuing the property as a nursing home as well as other alternative uses.

Assets Held for Sale and Discontinued Operations

Pursuant to the provisions of SFAS No. 144, Accounting for the Impairment or Disposal of Long-Lived Assets, the operating results of specified real estate assets that have been sold, or otherwise qualify as held for disposition (as defined by SFAS No. 144), are reflected as discontinued operations in the consolidated statements of operations for all periods presented.

Consolidation of Variable Interest Entities

Financial Accounting Standards Board Interpretation No. 46R, Consolidation of Variable Interest Entities, or FIN 46R, addresses the consolidation by business enterprises of variable interest entities ("VIEs"). FIN 46R provides criteria that, if met, would create a VIE, which would then be subject to consolidation. If an entity is determined to be a VIE, the party deemed to be the primary beneficiary is required to consolidate the VIE. The primary beneficiary is the party that has the most variability in expected gains or losses of the VIE.

In order to determine which party is the primary beneficiary, we must calculate the expected losses and expected residual returns of the VIE. The analysis requires a projection of expected cash flows and the assignment of probabilities to each possible cash flow outcome. Estimating expected cash flows and assigning probabilities of each outcome of the VIE requires us to use significant judgment. If assumptions used to estimate the expected cash flows prove to be inaccurate, our conclusions regarding which party is the VIE's primary beneficiary could be incorrect, resulting in us improperly consolidating the VIE when we are not the primary beneficiary or not consolidating the VIE when we are the primary beneficiary.

Results of Operations

The following is our discussion of the consolidated results of operations, financial position and liquidity and capital resources, which should be read in conjunction with our audited consolidated financial statements and accompanying notes.

Three Months Ended June 30, 2006 and 2005

Operating Revenues

Our operating revenues for the three months ended June 30, 2006 totaled \$31.1 million, an increase of \$5.8 million, over the same period in 2005. The \$5.8 million increase was primarily a result of new investments made throughout 2005 as well as scheduled contractual increases in rents. The increase in operating revenues from new investments was partially offset by a reduction in mortgage interest income and one-time contractual interest revenue associated with the payoff of a mortgage during the first quarter of 2005.

Detailed changes in operating revenues for the three months ended June 30, 2006 are as follows:

- · Rental income was \$29.0 million, an increase of \$6.5 million over the same period in 2005. The increase was due to new leases entered into throughout 2005, scheduled contractual increases in rents, and from consolidation of a VIE.
- · Mortgage interest income totaled \$1.2 million, a decrease of \$0.1 million over the same period in 2005. The decrease was primarily the result of normal amortization and a \$10 million loan payoff that occurred in the second quarter of 2006.
- Miscellaneous revenue was \$0.3 million, a decrease of \$0.8 million over the same period in 2005. The decrease was due to contractual revenue owed to us and received in the second quarter of 2005 resulting from a mortgage note prepayment that occurred in the first quarter of 2005.

Operating Expenses

Operating expenses for the three months ended June 30, 2006 totaled \$9.9 million, an increase of approximately \$0.9 million over the same period in 2005. The increase was primarily due to \$1.5 million of increased depreciation expense partially offset by a 2005 leasehold expiration expense.

Detailed changes in our operating expenses for the three months ended June 30, 2006 versus the same period in 2005 are as follows:

- · Our depreciation and amortization expense was \$7.5 million, compared to \$6.0 million for the same period in 2005. The increase is due to new investments placed throughout 2005 and the consolidation of a VIE in 2006.
- Our general and administrative expense, when excluding restricted stock amortization expense, was \$2.0 million, compared to \$1.8 million for the same period in 2005. The increase was primarily due to normal inflationary increases in goods and services.
- · In 2005, we recorded a \$0.1 million provision for uncollectible notes receivable.
- · In 2005, we recorded a \$0.8 million lease expiration accrual relating to disputed capital improvement requirements associated with a lease that expired June 30, 2005.

Other Income (Expense)

For the three months ended June 30, 2006, our total other net expenses were \$9.8 million as compared to \$10.8 million for the same period in 2005. The significant changes are as follows:

- Our interest expense, excluding amortization of deferred costs and refinancing related interest expenses, for the three months ended June 30, 2006 was \$9.4 million, compared to \$6.9 million for the same period 2005. The increase of \$2.5 million was primarily due to higher debt on our balance sheet versus the same period in 2005 and from consolidation of a VIE in 2006.
- During the three months ended June 30, 2005, we recorded a \$3.4 million provision for impairment on an equity security. In accordance with FASB Statement No. 115, *Accounting for Certain Investments in Debt and Equity Securities*, the \$3.4 million provision for impairment was to write-down our 760,000 share investment in Sun Healthcare Group, Inc.'s ("Sun") common stock to its then current fair market value.

Six Months Ended June 30, 2006 and 2005

Operating Revenues

Our operating revenues for the six months ended June 30, 2006 totaled \$61.8 million, an increase of \$9.4 million, over the same period in 2005. The \$9.4 million increase was primarily a result of new investments made throughout 2005 as well as scheduled contractual increases in rents. The increase in operating revenues from new investments was partially offset by a reduction in mortgage interest income and one-time contractual interest revenue associated with the payoff of a mortgage during the first quarter of 2005.

Detailed changes in operating revenues for the six months ended June 30, 2006 are as follows:

- Rental income was \$58.0 million, an increase of \$13.7 million over the same period in 2005. The increase was due to new leases entered into throughout 2005, scheduled contractual increases in rents, and the consolidation of a VIE in 2006.
- · Mortgage interest income totaled \$2.3 million, a decrease of \$0.9 million over the same period in 2005. The decrease was primarily the result of normal amortization, a \$60 million loan payoff that occurred in the first quarter of 2005 and a \$10 million loan payoff that occurred in the second quarter of 2006.
- · Miscellaneous revenue was \$0.4 million, a decrease of \$3.9 million over the same period in 2005. The decrease was due to contractual revenue owed to us and received in the second quarter of 2005 resulting from a mortgage note prepayment that occurred in the first quarter of 2005.

Operating Expenses

Operating expenses for the six months ended June 30, 2006 totaled \$19.7 million, an increase of approximately \$2.9 million over the same period in 2005. The increase was primarily due to \$3.3 million of increased depreciation expense partially offset by a 2005 leasehold termination expense.

Detailed changes in our operating expenses for the six months ended June 30, 2006 versus the same period in 2005 are as follows:

- · Our depreciation and amortization expense was \$15.1 million, compared to \$11.7 million for the same period in 2005. The increase is due to new investments placed throughout 2005 and the consolidation of a VIE in 2006.
- Our general and administrative expense, when excluding restricted stock amortization expense, was \$4.1 million, compared to \$3.7 million for the same period in 2005. The increase was primarily due to normal inflationary increases in goods and services.
- · In 2005, we recorded a \$0.1 million provision for uncollectible notes receivable.
- · In 2005, we recorded a \$0.8 million lease expiration accrual relating to disputed capital improvement requirements associated with a lease that expired June 30, 2005.

Other Income (Expense)

For the six months ended June 30, 2006, our total other net expenses were \$23.4 million as compared to \$18.0 million for the same period in 2005. The significant changes are as follows:

- · Our interest expense, excluding amortization of deferred costs and refinancing related interest expenses, for the six months ended June 30, 2006 was \$19.1 million, compared to \$13.7 million for the same period 2005. The increase of \$5.3 million was primarily due to higher debt on our balance sheet versus the same period in 2005 and from consolidation of a VIE in 2006.
- · For the six months ended June 30, 2006, we recorded a \$0.8 million non-cash charge associated with the redemption of the remaining 20.7% of our \$100 million aggregate principal amount of 6.95% unsecured notes due 2007 not otherwise tendered in 2005.
- · For the six months ended June 30, 2006, we recorded a one time, non-cash charge of approximately \$2.7 million relating to the write-off of deferred financing costs associated with the termination of our prior credit facility.
- During the six months ended June 30, 2005, we recorded a \$3.4 million provision for impairment of an equity security. In accordance with FASB Statement No. 115, *Accounting for Certain Investments in Debt and Equity Securities*, the \$3.4 million provision for impairment was to write-down our 760,000 share investment in Sun's common stock to its then current fair market value.

Loss from Discontinued Operations

Discontinued operations relate to properties we disposed of in 2006 or are currently held-for-sale and are accounted for as discontinued operations under SFAS No. 144. For the three and six months ended June 30, 2006, we sold two SNFs in California resulting in an accounting loss of approximately \$0.1 million

At June 30, 2006, we had one asset held for sale with a net book value of approximately \$0.2 million.

During the three months ended March 31, 2006, a \$0.1 million provision for impairment charge was recorded to reduce the carrying value to its sales price of one facility that was under contract to be sold that was subsequently sold during the second quarter of 2006.

In accordance with SFAS No. 144, the \$0.4 million realized net loss is reflected in our consolidated statements of operations as discontinued operations. See Note 9 - Discontinued Operations.

Funds From Operations

Our funds from operations available to common stockholders ("FFO"), for the three months ended June 30, 2006, was \$16.5 million, compared to \$8.1 million, for the same period in 2005.

We calculate and report FFO in accordance with the definition and interpretive guidelines issued by the National Association of Real Estate Investment Trusts ("NAREIT"), and, consequently, FFO is defined as net income available to common stockholders, adjusted for the effects of asset dispositions and certain non-cash items, primarily depreciation and amortization. We believe that FFO is an important supplemental measure of our operating performance. Because the historical cost accounting convention used for real estate assets requires depreciation (except on land), such accounting presentation implies that the value of real estate assets diminishes predictably over time, while real estate values instead have historically risen or fallen with market conditions. The term FFO was designed by the real estate industry to address this issue. FFO herein is not necessarily comparable to FFO of other real estate investment trusts ("REITs") that do not use the same definition or implementation guidelines or interpret the standards differently from us.

We use FFO as one of several criteria to measure operating performance of our business. We further believe that by excluding the effect of depreciation, amortization and gains or losses from sales of real estate, all of which are based on historical costs and which may be of limited relevance in evaluating current performance, FFO can facilitate comparisons of operating performance between periods and between other REITs. We offer this measure to assist the users of our financial statements in evaluating our financial performance under GAAP, and FFO should not be considered a measure of liquidity, an alternative to net income or an indicator of any other performance measure determined in accordance with GAAP. Investors and potential investors in our securities should not rely on this measure as a substitute for any GAAP measure, including net income.

In February 2004, NAREIT informed its member companies that it was adopting the position of the Securities and Exchange Commission ("SEC") with respect to asset impairment charges and would no longer recommend that impairment write-downs be excluded from FFO. In the tables included in this disclosure, we have applied this interpretation and have not excluded asset impairment charges in calculating our FFO. As a result, our FFO may not be comparable to similar measures reported in previous disclosures. According to NAREIT, there is inconsistency among NAREIT member companies as to the adoption of this interpretation of FFO. Therefore, a comparison of our FFO results to another company's FFO results may not be meaningful.

The following table presents our FFO results reflecting the impact of asset impairment charges (the SEC's interpretation) for the three and six months ended June 30, 2006 and 2005:

	Three Months Ended			Six Months Ended June 30,					
	June 30,								
		2006		2005		2006		2005	
				(in thou	sands)				
Net income available to common	\$	8,780	\$	(2,620)	\$	13,180	\$	3,125	
Add back loss from real estate dispositions		133		4,165		381		4,202	
Sub-total Elimination of non-cash items included in net income:		8,913		1,545		13,561		7,327	
Depreciation and amortization		7,542		6,540		15,069		12,793	
Funds from operations available to common stockholders	\$	16,455	\$	8,085	\$	28,630	\$	20,120	

Taxes

As a qualified REIT, as long as we distribute 100% of our taxable income, we will not be subject to Federal income taxes on our income, and no provisions for Federal income taxes have been made. We are permitted to own up to 100% of a "taxable REIT subsidiary" ("TRS"). Currently, we have two TRSs that are taxable as corporations and that pay federal, state and local income tax on their net income at the applicable corporate rates.

Portfolio Developments, New Investments and Recent Developments

Below is a brief description, by third-party operator, of our re-leasing, restructuring or new investment transactions that occurred during the six months ended June 30, 2006.

Hickory Creek Healthcare Foundation, Inc.

On June 16, 2006, we received approximately \$10 million in proceeds on a mortgage loan payoff. We held mortgages on 15 facilities located in Indiana, representing 619 beds.

Haven Eldercare, LLC

- During the three months ending March 31, 2006, Haven Eldercare, LLC ("Haven"), an existing operator of ours, entered into a \$39 million first mortgage loan with General Electric Capital Corporation ("GE Loan"). Haven used the \$39 million of proceeds to partially repay on a \$62 million mortgage it has with us. Simultaneously, we subordinated the payment of our remaining \$23 million on the mortgage note, due in October 2012, to that of the GE Loan. As a result of this transaction, the interest rate on our remaining mortgage note to Haven rose from 10% to approximately 15%, with annual escalators.
- In conjunction with the above transactions and the application of FIN 46R, we consolidated the financial statements and related real estate of this Haven entity into our financial statements. The consolidation resulted in the following changes to our consolidated balance sheet as of June 30, 2006: (1) an increase in total gross investments of \$39.0 million; (2) an increase in accumulated depreciation of \$0.8 million; (3) an increase in other long-term borrowings of \$39.0 million; and (4) a reduction of \$0.8 million in cumulative net earnings for the six months ended June 30, 2006 due to the increased depreciation expense. General Electric Capital Corporation and Haven's other creditors do not have recourse to our assets. Our results of operations reflect the effects of the consolidation of this entity, which is being accounted for similarly to our other purchase-leaseback transactions.

Assets Sold

- · On June 30, 2006, we sold two SNFs in California resulting in an accounting loss of approximately \$0.1 million.
- · On March 31, 2006, we sold a SNF in Illinois resulting in an accounting loss of approximately \$0.2 million.

Held for Sale

- · We had one asset held for sale as of June 30, 2006 with a net book value of approximately \$0.2 million.
- · During the three months ended March 31, 2006, a \$0.1 million provision for impairment charge was recorded to reduce the carrying value to its sales price of one facility that was under contract to be sold, which was subsequently sold during the second quarter of 2006.

The partial expiration of certain Medicare rate increases has had an adverse impact on the revenues of the operators of nursing home facilities and has negatively impacted some operators' ability to satisfy their monthly lease or debt payment to us. In several instances, we hold security deposits that can be applied in the event of lease and loan defaults, subject to applicable limitations under bankruptcy law with respect to operators seeking protection under Chapter 11 of the Bankruptcy Code.

Liquidity and Capital Resources

At June 30, 2006, we had total assets of \$984.4 million, stockholders' equity of \$435.3 million and debt of \$526.5 million, which represents approximately 54.7% of our total capitalization.

The following table shows the amounts due in connection with the contractual obligations described below as of June 30, 2006.

	 Payments due by period								
	 Total	_	Less than 1 year	_	1-3 years (In thousands	_	3-5 years	_	More than 5 years
Long-term debt (1)	\$ 526,800	\$	390		850	•	960	\$	524,600
Other long-term liabilities	 616		231	_	385	_	-	_	
Total	\$ 527,416	\$	621	\$	1,235	\$	960	\$	524,600

⁽¹⁾ The \$526.8 million includes \$310 million aggregate principal amount of 7% Senior Notes due 2014, \$175 million aggregate principal amount of 7% Senior Notes due 2016 and Haven's \$39 million first mortgage with General Electric Capital Corporation that expires in 2012. As of June 30, 2006, there are no outstanding borrowings under the new \$200 million revolving senior secured credit facility that matures in March 2010.

Financing Activities and Borrowing Arrangements

Bank Credit Agreements

At June 30, 2006, we had no outstanding borrowings under our \$200 million revolving senior secured credit facility (the "New Credit Facility"); however, \$2.9 million was utilized for the issuance of letters of credit, leaving availability of \$197.1 million. The New Credit Facility, entered into on March 31, 2006, is being provided by Bank of America, N.A., as Administrative Agent, Deutsche Bank Trust

Company Americas, UBS Securities LLC, General Electric Capital Corporation, LaSalle Bank N.A., and Citicorp North America, Inc. and will be used for acquisitions and general corporate purposes.

The New Credit Facility replaced our previous \$200 million senior secured credit facility (the "Prior Credit Facility") that was terminated on March 31, 2006. We will realize a 125 basis point savings on LIBOR-based loans under the New Credit Facility, as compared to LIBOR-based loans under our Prior Credit Facility. The New Credit Facility matures on March 31, 2010, and includes an "accordion feature" that permits us to expand our borrowing capacity to \$300 million during our first two years.

For the three-month period ending March 31, 2006, we recorded a one-time, non-cash charge of approximately \$2.7 million relating to the write-off of deferred financing costs associated with the termination of our Prior Credit Facility.

Our long-term borrowings require us to meet certain property level financial covenants and corporate financial covenants, including prescribed leverage, fixed charge coverage, minimum net worth, limitations on additional indebtedness and limitations on dividend payouts. As of June 30, 2006, we were in compliance with all property level and corporate financial covenants.

\$100 Million Aggregate Principal Amount of 6.95% Unsecured Notes Tender and Redemption

On December 16, 2005, we initiated a tender offer and consent solicitation for all of our outstanding \$100 million aggregate principal amount 6.95% notes due 2007 (the "2007 Notes"). On December 30, 2005, we accepted for purchase the 79.3% of the aggregate principal amount of the 2007 Notes outstanding that were tendered. On December 30, 2005, our Board of Directors also authorized the redemption of all outstanding 2007 Notes that were not otherwise tendered. On December 30, 2005, upon our irrevocable funding of the full redemption price for the 2007 Notes and certain other acts required by the Indenture governing the 2007 Notes, the Trustee of the 2007 Notes certified in writing to us (the "Certificate of Satisfaction and Discharge") that the Indenture was satisfied and discharged as of December 30, 2005, except for certain administrative provisions. In accordance with FASB Statement No. 140, *Accounting for Transfers and Servicing of Financial Assets and Extinguishment of Liabilities*, ("FAS 140") we removed 79.3% of the aggregate principal amount of the 2007 Notes, which were tendered in our tender offer and consent solicitation, and the corresponding portion of the funds held in trust by the Trustee to pay the tender price from our balance sheet and recognized \$2.8 million of additional interest expense associated with the tender offer. On January 18, 2006, we completed the redemption of the remaining 2007 Notes not otherwise tendered. Accordingly, we reduced other assets, representing the funds deposited with the Trustee, and unsecured borrowings by \$21 million. In connection with the redemption and in accordance with FAS 140, we recognized \$0.8 million of additional interest expense in the first quarter of 2006. As of January 18, 2006, none of the 2007 Notes remained outstanding.

Other Long-Term Borrowings

During the three months ended March 31, 2006, Haven used the \$39 million of proceeds from the GE Loan to partially repay a portion of a \$62 million mortgage it has with us. Simultaneously, we subordinated the payment of its remaining \$23 million on the mortgage note to that of the GE Loan. In conjunction with the above transactions and the application of FIN 46R, we consolidated the financial statements of this Haven entity into our financial statements, which contained the long-term borrowings with General Electric Capital Corporation of \$39.0 million. The loan has an interest rate of approximately seven percent and is due in 2012. The lender of the \$39.0 million does not have recourse to our assets. See Note - 2 Properties; Leased Property.

Dividends

In order to qualify as a REIT, we are required to distribute dividends (other than capital gain dividends) to our stockholders in an amount at least equal to (A) the sum of (i) 90% of our "REIT taxable income" (computed without regard to the dividends paid deduction and our net capital gain), and (ii) 90% of the net income (after tax), if any, from foreclosure property, minus (B) the sum of certain items of non-cash income. In addition, if we dispose of any built-in gain asset during a recognition period, we will be required to distribute at least 90% of the built-in gain (after tax), if any, recognized on the disposition of such asset. Such distributions must be paid in the taxable year to which they relate, or in the following taxable year if declared before we timely file our tax return for such year and paid on or before the first regular dividend payment after such declaration. In addition, such distributions are required to be made pro rata, with no preference to any share of stock as compared with other shares of the same class, and with no preference to one class of stock as compared with another class except to the extent that such class is entitled to such a preference. To the extent that we do not distribute all of our net capital gain or do distribute at least 90%, but less than 100% of our "REIT taxable income," as adjusted, we will be subject to tax thereon at regular ordinary and capital gain corporate tax rates. In addition, our \$200 million New Credit Facility has certain financial covenants that limit the distribution of dividends paid during a fiscal quarter to no more than 95% of our aggregate cumulative funds from operations ("FFO") as defined in the loan agreement governing the New Credit Facility (the "Loan Agreement"), unless a greater distribution is required to maintain REIT status. The Loan Agreement defines FFO as net income (or loss) plus depreciation and amortization and shall be adjusted for charges related to: (i) restructuring our debt; (ii) redemption of preferred stock; (iii)

Common Dividends

On July 17, 2006, the Board of Directors announced a common stock dividend of \$0.24 per share to be paid August 15, 2006 to common stockholders of record on July 31, 2006.

On April 18, 2006, the Board of Directors declared a common stock dividend of \$0.24 per share, an increase of \$0.01 per common share compared to the prior quarter. The common dividend was paid May 15, 2006 to common stockholders of record on April 28, 2006.

On January 17, 2006, the Board of Directors declared a common stock dividend of \$0.23 per share, an increase of \$0.01 per common share compared to the prior quarter. The common stock dividend was paid February 15, 2006 to common stockholders of record on January 31, 2006.

Series D Preferred Dividends

On July 17, 2006, the Board of Directors declared the regular quarterly dividends for the 8.375% Series D Cumulative Redeemable Preferred Stock ("Series D Preferred Stock") to stockholders of record on July 31, 2006. The stockholders of record of the Series D Preferred Stock on July 31, 2006 will be paid dividends in the amount of \$0.52344 per preferred share on August 15, 2006. The liquidation preference for our Series D Preferred Stock is \$25.00 per share. Regular quarterly preferred dividends for the Series D Preferred Stock represent dividends for the period May 1, 2006 through July 31, 2006.

Liquidity

We believe our liquidity and various sources of available capital, including cash from operations, our existing availability under our New Credit Facility and expected proceeds from mortgage payoffs are more than adequate to finance operations, meet recurring debt service requirements and fund future investments through the next twelve months.

We regularly review our liquidity needs, the adequacy of cash flow from operations, and other expected liquidity sources to meet these needs. We believe our principal short-term liquidity needs are to fund:

- normal recurring expenses;
- · debt service payments;
- preferred stock dividends;
- · common stock dividends; and
- · growth through acquisitions of additional properties.

The primary source of liquidity is our cash flows from operations. Operating cash flows have historically been determined by: (i) the number of facilities we lease or have mortgages on; (ii) rental and mortgage rates; (iii) our debt service obligations; and (iv) general and administrative expenses. The timing, source and amount of cash flows provided by financing activities and used in investing activities are sensitive to the capital markets environment, especially to changes in interest rates. Changes in the capital markets environment may impact the availability of cost-effective capital and affect our plans for acquisition and disposition activity.

Cash and cash equivalents totaled \$14.1 million as of June 30, 2006, an increase of \$10.1 million as compared to the balance at December 31, 2005. The following is a discussion of changes in cash and cash equivalents due to operating, investing and financing activities, which are presented in our Consolidated Statements of Cash Flows.

<u>Operating Activities</u> - Net cash flow from operating activities generated \$42.7 million for the six months ended June 30, 2006, as compared to \$34.4 million for the same period in 2005. The \$8.3 million increase is due primarily to: (i) incremental revenue associated with acquisitions completed throughout 2005 and (ii) normal working capital fluctuations during the period.

<u>Investing Activities</u> - Net cash flow from investing activities was an inflow of \$4.3 million for the six months ended June 30, 2006, as compared to an outflow of \$53.9 million for the same period in 2005. The \$58.2 million change in investing cash outflow was primarily due to: (i) an outflow of \$120.7 million for acquisitions completed during the six months of 2005 as compared to the same period in 2006, (ii) a \$60.0 million mortgage payoff in 2005 as compared to a \$10.0 million mortgage payoff in 2006 and (iii) \$0.7 million collected from the sale of real estate for the six months ended June 30, 2006 as compared to \$12.3 million in 2005.

Financing Activities - Net cash flow from financing activities was an outflow of \$36.9 million for the six months ended June 30, 2006 as compared to an inflow of \$7.9 million for the same period in 2005. The increase in financing cash outflow of \$44.8 million was primarily a result of net repayments of \$58.0 million on our New Credit Facility in 2006 versus net borrowings of \$86.5 million on our Prior Credit Facility in 2005. The financial cash outflow was partially offset by \$39.0 million of proceeds received in a Haven transaction, \$50.0 million redemption of the outstanding shares of Series B preferred stock in 2005 and \$16.6 million of incremental optional cash purchases of our common stock through our dividend reinvestment and common stock purchase plan for the six months ended June 30, 2006 as compared to the same time period in 2005.

Item 3 - Quantitative and Qualitative Disclosures about Market Risk

There were no material changes in our market risk during the three months ended June 30, 2006. For additional information, refer to Item 7A as presented in our annual report on Form 10-K for the year ended December 31, 2005.

Item 4 - Controls and Procedures

Evaluation of Disclosure Controls and Procedures

Our principal executive officer and principal financial officer are responsible for establishing and maintaining disclosure controls and procedures as defined in the rules promulgated under the Securities Exchange Act of 1934, as amended (the "Exchange Act"). We evaluated the effectiveness of the design and operation of our disclosure controls and procedures as of June 30, 2006, and based on that evaluation, our principal executive officer and principal financial officer have concluded that these controls and procedures were effective as of June 30, 2006.

Disclosure controls and procedures are the controls and other procedures designed to ensure that information that we are required to disclose in our reports under the Exchange Act is recorded, processed, summarized and reported within the time periods required. Disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information we are required to disclose in the reports that we file or submit under the Exchange Act is accumulated and communicated to our management, including our principal executive officer and principal financial officer, as appropriate, to allow timely decisions regarding required disclosure.

Changes in Internal Control Over Financial Reporting

No changes in our internal control over financial reporting were identified as having occurred in the three months ended June 30, 2006 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II - OTHER INFORMATION

Item 1 - Legal Proceedings

See Note 8 - Litigation to the Consolidated Financial Statements in PART I, Item 1 hereto, which is hereby incorporated by reference in response to this item.

Item 1A- Risk Factors

We filed our Annual Report on Form 10-K for the year ended December 31, 2005 with the Securities and Exchange Commission on February 17, 2006, which sets forth our risk factors in Item 1A therein. We have not experienced any material changes from the risk factors previously described therein.

Item 4 - Submission of Matters to a Vote of Security Holders

Our annual meeting of stockholders (the "Annual Meeting") was held on May 25, 2006. Of the total number of common shares outstanding on April 21, 2006, a total of 57,992,789 were represented in person or by proxy at the Annual Meeting. Results of votes with respect to proposals submitted at the Annual Meeting are set forth below.

(a) To elect two nominees to serve as directors and to hold office until the next annual meeting of stockholders or until their successors have been elected and qualified. Our stockholders voted to elect both nominees to serve as directors. Votes recorded, by nominee, were as follows:

Nominee	For	Against/Withheld
Bernard J. Korman	51,060,573	649,442
Thomas F. Franke	51,056,153	653,862

(b) To consider and vote upon a proposal to ratify the selection of Ernst & Young LLP as our independent auditor for the fiscal year 2006:

For	Against	Abstain
51,555,820	62,564	91,630

Item 5 - Other Information

We entered into a contract of sale (the "Litchfield Contract"), dated as of May 5, 2006, between Laramie Associates, LLC, Casper Associates, LLC, North 12th Street Associates, LLC, North Union Boulevard Associates, LLC, Jones Avenue Associates, LLC, Litchfield Investment Company, L.L.C., Ustick Road Associates, LLC, West 24th Street Associates, LLC, North Third Street Associates, LLC, Midwestern Parkway Associates, LLC, North Francis Street Associates, LLC, West Nash Street Associates, LLC (as sellers) and OHI Asset (LA), LLC, NRS Ventures, L.L.C. and OHI Asset (CO), LLC (as buyers).

The Litchfield Contract provides for the purchase of thirty skilled nursing facilities and one independent living center from Litchfield Companies for approximately \$171 million (the "Litchfield Transaction"). The facilities total 3,847 beds and are located in the states of Colorado (5), Florida (7),

Idaho (1), Louisiana (13), and Texas (5). The facilities were subject to master leases with three national healthcare providers, which are our existing tenants. The tenants are Home Quality Management, Inc., Nexion Health, Inc., and Peak Medical Corporation, which was acquired by Sun Healthcare Group, Inc. ("Sun") in December of 2005. The Litchfield Contract provides for customary representations, warranties and indemnification provisions. The Litchfield Contract is attached to this Form 10-Q as Exhibit 10.1 and is incorporated herein by reference. The total incremental annualized rent from the 31 facility transaction is approximately \$17.1 million.

We closed the Litchfield Transaction on August 1, 2006. We used a combination of cash on hand and \$150 million of borrowings under our \$200 million revolving senior secured credit facility (the "New Credit Facility"), drawn on August 1, 2006, to finance the Litchfield transaction.

The New Credit Facility, entered into on March 31, 2006, is being provided by Bank of America, N.A., as Administrative Agent, Deutsche Bank Trust Company Americas, UBS Securities LLC, General Electric Capital Corporation, LaSalle Bank N.A., and Citicorp North America, Inc. and will be used for acquisitions and general corporate purposes. The New Credit Facility matures on March 31, 2010, and includes an "accordion feature" that permits us to expand our borrowing capacity to \$300 million during our first two years.

Item 6 - Exhibits

Exhibit No.	Description
10.1	Contract of sale, dated as of May 5, 2006, between Laramie Associates, LLC, Casper Associates, LLC, North 12 th Street Associates, LLC, North Union Boulevard Associates, LLC, Jones Avenue Associates, LLC, Litchfield Investment Company, L.L.C., Ustick Road Associates, LLC, West 24 th Street Associates, LLC, North Third Street Associates, LLC, Midwestern Parkway Associates, LLC, North Francis Street Associates, LLC, West Nash Street Associates, LLC (as sellers) and OHI Asset (LA), LLC, NRS Ventures, L.L.C. and OHI Asset (CO), LLC (as buyers).
31.1	Rule 13a-14(a)/15d-14(a) Certification of the Chief Executive Officer.
31.2	Rule 13a-14(a)/15d-14(a) Certification of the Chief Financial Officer.
32.1	Section 1350 Certification of the Chief Executive Officer.
32.2	Section 1350 Certification of the Chief Financial Officer.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

OMEGA HEALTHCARE INVESTORS, INC. Registrant

Date: August 4, 2006 By: <u>/S/ C. TAYLOR PICKETT</u>

C. Taylor Pickett Chief Executive Officer

Date: August 4, 2006 By: <u>/S/ ROBERT O. STEPHENSON</u>

Robert O. Stephenson Chief Financial Officer

CONTRACT OF SALE

BETWEEN

THE ENTITIES LISTED ON EXHIBIT A HERETO

AND

THE ENTITIES LISTED ON EXHIBIT B HERETO

DATED AS OF MAY 5, 2006

CONTRACT OF SALE

THIS CONTRACT OF SALE (this "Agreement"), dated as of May 5, 2006, is between THE ENTITIES LISTED ON EXHIBIT A HERETO (collectively, "Sellers") and THE ENTITIES LISTED ON EXHIBIT B HERETO (collectively, "Buyers").

- 1. Sale and Purchase of Properties.
 - 1.1. Subject to the conditions and upon the terms of this Agreement, Sellers shall sell and transfer to Buyers, and Buyers shall purchase from Sellers, Sellers' rights, title and interests in and to all of the Real Estate, Buildings and Personalty constituting the Facilities described on Exhibit C hereto (individually, a "Property" and, collectively, the "Properties").
 - 1.2. For purposes of this Agreement, the following capitalized terms have the following meanings:
 - 1.2.1. "Agreement" as defined in the first paragraph hereof and including all Exhibits and Schedules attached hereto and made a part hereof.
 - 1.2.2. "Bills of Sale" means, collectively, the Bills of Sale, each dated as of the Closing Date, to be provided from Sellers to Buyers, transferring the Personalty from Sellers to Buyers.
 - 1.2.3. "Buildings" as defined in Section 1.2.54 hereof.
 - 1.2.4. "Buyers" as defined in the first paragraph hereof.
 - 1.2.5. "Buyers Indemnified Parties" as defined in Section 15.1 hereof.
 - 1.2.6. "Buyers' Knowledge" means the actual knowledge of C. Taylor Pickett and Daniel J. Booth.
 - 1.2.7. "Buyers Transaction Documents" means, collectively, this Agreement, the Bills of Sale, the Peak Assignment Agreement, the Peak Guaranty Assignment, the L/C Agreement, and all other documents and agreements now or hereafter to be executed and delivered by Buyers pursuant to this Agreement.
 - 1.2.8. "Claims" as defined in Section 15.1 hereof.
 - 1.2.9. "Closing" as defined in Section 9.1.1 hereof.
 - 1.2.10. "Closing Date" as defined in Section 9.1.1 hereof.
 - 1.2.11. "Commitments" as defined in Section 7.3.1 hereof.
 - 1.2.12. "Deeds" means, collectively, the special warranty deeds or the equivalent versions of special warranty deeds as may be utilized under applicable

state law for any Properties, each dated as of the Closing Date, to be provided from Sellers to Buyers, transferring the Real Estate from Sellers to Buyers.

- 1.2.13. "Environmental Claim" means the violation or alleged violation of an Environmental Law or the existence, presence or Release of any Hazardous Substances in connection with or based on any occurrence, event or condition at or relating to the use, possession, operation or management of the Properties.
- 1.2.14. "Environmental Laws" means the rules and regulations of the Environmental Protection Agency and all applicable rules and regulations of federal, state and local laws, including statutes, regulations, ordinances, codes, rules, as amended, relating to the discharge of air pollutants, water pollutants or process waste water or Hazardous Substances or toxic substances including, but not limited to, the Federal Toxic Substances Act, the Federal Solid Waste Disposal Act, the Federal Clean Air Act, the Federal Clean Water Act, the Federal Resource Conservation and Recovery Act of 1976, the Federal Comprehensive Environmental Response, Compensation and Liability Act of 1980, each as amended, regulations of the Environmental Protection Agency, regulations of the Nuclear Regulatory Agency, and regulations of any state department of natural resources or state environmental protection agency in effect at the Closing.
- 1.2.15. "Facility" or "Facilities" means the skilled nursing facilities and independent living facility listed as Facilities on Exhibit C hereto.
- 1.2.16. "Government Entity" means the United States of America or any state or other political subdivision thereof, or any entity exercising executive, legislative, judicial, regulatory or administrative functions of government.
 - 1.2.17. "Guarantors" means, collectively, each Guarantor as defined in the Member Guarantees.
- 1.2.18. "Hazardous Substances" means any wastes, substances, or materials (whether solids, liquids or gases) that are deemed hazardous, toxic, pollutants or contaminants, including without limitation, substances defined as "hazardous wastes," "hazardous substances," "toxic substances," "radioactive materials," or other similar designations in, or otherwise subject to regulation under, any Environmental Laws. "Hazardous Substances" includes polychlorinated biphenyls (PCBs), asbestos, lead-based paints, infectious wastes, radioactive materials and wastes and petroleum and petroleum products.
 - 1.2.19. "HQM" means Home Quality Management, Inc.
- 1.2.20. "HQM Master Lease" means the Master Lease Agreement, dated as of October 31, 2002, between LIC and the affiliate Tenants of HQM described as Tenant therein.

- 1.2.21. "HQM-Omega Master Lease" means the Amended and Restated Master Lease, dated as of the Closing Date, between NRS and the affiliate Tenants of HQM.
- 1.2.22. "HQM Option Agreement" means the Purchase Option Agreement, dated as of October 31, 2002, between LIC and the affiliate Tenants of HQM described as Buyer therein.
- 1.2.23. "HQM Termination Agreement" means the Termination Agreement, dated as of the Closing Date, between LIC and the affiliate Tenants of HQM.
 - 1.2.24. "Inspection Period" as defined in Section 8.1.3(a) hereof.
 - 1.2.25. "Land" as defined in Section 1.2.54.
- 1.2.26. "L/C Agreement" means the Letter of Credit Agreement, dated as of the Closing Date, between Sellers and Buyers.
- 1.2.27. "Letter of Credit" means the Letter of Credit, as defined in the L/C Agreement, in the amount of Three Million Dollars (\$3,000,000), caused to be provided by Sellers to Buyers.
 - 1.2.28. "LIC" means Litchfield Investment Company, L.L.C.
- 1.2.29. "Litchfield Master Leases" means, collectively, the Nexion Master Lease, the HQM Master Lease and the Peak Master Lease.
- 1.2.30. "Material Adverse Change" means any changes that materially and adversely impacts any of the businesses, operations or financial condition of the Tenants or the Facilities, whether taken as a whole or as a group under the Nexion Master Lease, the HQM Master Lease or the Peak Master Lease, which are directly caused by any Acts of God, acts of public enemies or terrorists, war, other military conflicts, blockades, insurrections, riots, epidemics, quarantine restrictions, landslides, lightning, earthquake, fires, conflagration, storms, floods, washouts, civil disturbances, or any actions of any Governmental Entity.
 - 1.2.31. "McGee" means Michael S. McGee.
 - 1.2.32. "Nexion" means Nexion Health, Inc.
- 1.2.33. "Nexion Master Lease" means the Master Lease Agreement, dated as of November 1, 2002, between the Sellers described as Landlord therein and the affiliate Tenants of Nexion described as Tenant therein.
- 1.2.34. "Nexion-Omega Master Lease" means the Amended and Restated Master Lease, dated as of the Closing Date, between OHI-LA and the affiliate Tenants of Nexion.

- 1.2.35. "Nexion Termination Agreement" means the Termination Agreement, dated as of the Closing Date, between the applicable Sellers and the affiliate Tenants of Nexion.
 - 1.2.36. "NRS" means NRS Ventures, L.L.C.
 - 1.2.37. "OHI-CO" means OHI Asset (CO), LLC.
 - 1.2.38. "OHI-LA" means OHI Asset (LA), LLC.
 - 1.2.39. "Omega" means Omega Healthcare Investors, Inc., a Maryland corporation.
- 1.2.40. "Omega Master Leases" means, collectively, the Nexion-Omega Master Lease and the HQM-Omega Master Lease.
- 1.2.41. "Peak Assignment Agreement" means the Assignment and Assumption Agreement with respect to the Peak Master Lease, dated as of the Closing Date, between OHI-CO and the applicable Sellers.
- 1.2.42. "Peak First Refusal Agreement" means the Right of First Refusal Agreement, dated as of October 31, 2002, between the applicable Sellers described as Seller therein and Peak No. 2.
- 1.2.43. "Peak Guaranty" means the Guaranty, entered into as of October 31, 2002, by Peak Medical for the benefit of the applicable Sellers described as Landlord therein.
- 1.2.44. "Peak Guaranty Assignment" means the Assignment Agreement with respect to the Peak Guaranty, dated as of the Closing Date, between OHI-CO and the applicable Sellers, with the written consent thereto from Peak Medical.
- 1.2.45. "Peak Master Lease" means the Master Lease Agreement, dated as of October 31, 2002, between the Sellers described as Landlord therein and Peak No. 2.
 - 1.2.46. "Peak Medical" means Peak Medical Corporation.
 - 1.2.47. "Peak No. 2" means Peak Medical Colorado No. 2, Inc.
 - 1.2.48. "Peak Properties" as defined in Section 3 hereof.
 - 1.2.49. "Permitted Exceptions" as defined in Section 7.1 hereof.
- 1.2.50. "Personalty" means, collectively, all fixtures, machinery, furniture, equipment, beds, and other tangible personal property owned by Sellers that is attached to, located at or used in the physical occupancy or operation of the Facilities and the Properties.

- 1.2.51. "Property" or "Properties" as defined in Section 1.1 hereof.
- 1.2.52. "Property Licenses" as defined in Section 6.1.5 hereof.
- 1.2.53. "Purchase Price" as defined in Section 5.1 hereof.
- 1.2.54. "Real Estate" or "Premises" means, collectively, all of the land situated at the addresses listed on Exhibit C hereto and more particularly described in the legal descriptions attached hereto as Exhibit D (collectively, the "Land"), together with and including Sellers' rights, title and interests in and to all of the rights, privileges, easements, transferable consents, authorizations, variances or waivers, licenses, permits and approvals, hereditaments and other appurtenances relating to the Land, and together with Sellers' rights, title and interests in and to all of the buildings, improvements and fixtures now or hereafter erected or located on the Land (collectively, the "Buildings").
- 1.2.55. "Release" means any release, spill, emission, leaking, pumping, injection, deposit, disposal, discharge, dispersal, leaching, or migration into the environment, including the movement of any Hazardous Substance or other substance through or in the air, soil, surface water, groundwater, or property.
- 1.2.56. "Relevant Date" means, with respect to any Material Adverse Change impacting the business, operations or financial condition of the Tenants or the Facilities, July 27, 2006.
- 1.2.57. "Remedial Action" means any action required under any Environmental Laws to (a) clean up, remove, treat, or in any other way address any Hazardous Substance or other substance in the environment, (b) prevent the Release or threat of Release, or minimize the further Release of any Hazardous Substance or other substance so it does not migrate or endanger or threaten to endanger public health or welfare or the environment, or (c) perform pre-remedial studies and investigations and post-remedial monitoring and care.
 - 1.2.58. "Rosen" means Eugene H. Rosen.
 - 1.2.59. "Sellers" as defined in the first paragraph hereof.
 - 1.2.60. "Sellers Indemnified Parties" as defined in Section 15.3 hereof.
- 1.2.61. "Sellers Transaction Documents" means, collectively, this Agreement, the Deeds, the Bills of Sale, the Peak Assignment Agreement, the Peak Guaranty Assignment, the L/C Agreement, and all other documents and agreements now or hereafter to be executed and delivered by Sellers pursuant to this Agreement; provided, however, that the HQM Termination Agreement and the Nexion Termination Agreement shall not be included as Sellers Transaction Documents.
 - 1.2.62. "Sellers' Office" means the notice address listed for Sellers in Section 14.17 hereof.

	1.2.63.	"Sellers' Knowledge" means the actual knowledge of Rosen, Weinstein and McGee.
	1.2.64.	"Surveys" means the existing land surveys with respect to the Properties.
	1.2.65.	"Survival Termination Date" as defined in Section 17 hereof.
<u>C</u> hereto.	1.2.66.	"Tenants" means the affiliates of Nexion, the affiliates of HQM, and Peak, which are listed as Tenants on Exhibit
	1.2.67.	"Title Company" as defined in Section 5.1 hereof.
Documents.	1.2.68.	"Transaction Documents" means, collectively, the Buyers Transaction Documents and the Sellers Transaction
	1.2.69.	"Weinstein" means Bruce Weinstein.

- 2. <u>Delivery of Omega Master Leases, Nexion Termination Agreement, HQM Termination Agreement, Peak Assignment Agreement, Peak Guaranty Assignment, L/C Agreement and Exhibits and Schedules to this Agreement.</u> On or before May 31, 2006, (a) Buyers shall deliver to Sellers the fully executed Omega Master Leases, (b) Sellers shall deliver to Buyers the fully executed HQM Termination Agreement and Nexion Termination Agreement, (c) the applicable Sellers and OHI-CO shall execute and deliver the Peak Assignment Agreement, (d) the applicable Sellers, OHI-CO and Peak Medical shall execute and deliver the Peak Guaranty Assignment, and (e) Sellers and Buyers shall execute and deliver the L/C Agreement. On or before May 15, 2006, Sellers shall deliver to Buyers Exhibits C and D and Schedules 6.1.2, 6.1.3, 6.1.6, 6.1.7 and 15.1(c) hereto.
- 3. Peak First Refusal Agreement. On or before May 25, 2006, the applicable Sellers shall provide written evidence to OHI-CO that they have complied with their obligations under the Peak First Refusal Agreement, resulting in Peak No. 2 exercising or failing to exercise its right to acquire those Properties covered by the Peak First Refusal Agreement (collectively, the "Peak Properties"). If Peak No. 2 exercises its right under the Peak First Refusal Agreement, then the Peak Properties shall be removed from being subject to sale and purchase under this Agreement and the Purchase Price shall be reduced by the amount associated with the Peak Properties on Exhibit C hereto; provided, however, if Peak No. 2 does not consummate its acquisition of the Peak Properties at the Closing, then the Closing shall be postponed and this Agreement shall be automatically extended for up to thirty (30) days so that Buyers shall acquire the Properties with the Peak Properties included and the Purchase Price shall be increased by the amount associated with the Peak Properties on Exhibit C hereto. It is expressly understood by the parties hereto that at the Closing, Sellers will sell all (and in no event some) of the Properties to Buyers and, if applicable, Peak No. 2.
- 4. <u>HQM Option Agreement</u>. The HQM Termination Agreement shall include provisions for the termination of the HQM Option Agreement.
- 5. <u>Purchase Price; Disbursement of Purchase Price and Closing Documents at Closing.</u>

- 5.1. <u>Purchase Price</u>. The total purchase price payable by Buyers to Sellers for the Properties shall be the sum of One Hundred Seventy Million Dollars (\$170,000,000) (the "Purchase Price"), which Purchase Price shall be allocated among the Properties in accordance with Exhibit C hereto. At the Closing, an amount equal to the Purchase Price shall be deposited by Buyers with First American Title Insurance Company of New York (the "Title Company") by wire transfer of same day funds.
- 5.2. <u>Disbursement of Purchase Price and Closing Documents at Closing</u>. Upon written notice from Sellers and Buyers to the Title Company that all conditions to Closing set forth in this Agreement have been satisfied or, as to any condition not satisfied, waived in writing by the party intended to be benefited thereby, on the Closing Date the Title Company shall distribute the following documents and funds in the following manner:
 - 5.2.1. cause the Deeds delivered by Sellers pursuant to Section 9.2.1 hereof to be recorded in the official records of the applicable counties where the Properties are located;
 - 5.2.2. cause the mortgage discharges and UCC-3 Termination Statements delivered by Sellers pursuant to Section 9.2.2 hereof to be recorded in the official records of the applicable counties and states where the Properties are located;
 - 5.2.3. deliver to Buyers all documents that are required to be delivered by Sellers to Buyers pursuant to Section 9.2 hereof: and
 - 5.2.4. deliver to Sellers (a) all documents that are required to be delivered by Buyers to Sellers pursuant to Section 9.3 hereof and (b) the Purchase Price and such other funds, if any, as may be due to Sellers by reason of credits under this Agreement, less the items chargeable to Sellers under this Agreement.

6. Representations and Warranties.

- 6.1. <u>Representations and Warranties by Sellers</u>. Sellers each represent and warrant to Buyers as follows:
- 6.1.1. Organization. Sellers are each a limited liability company duly organized and validly existing under the laws of the state noted for each of Sellers on Exhibit A hereto.
- 6.1.2. Power and Authority. Except as set forth on Schedule 6.1.2 hereto, Sellers have the right, power, and authority to execute and deliver this Agreement and the other Sellers Transaction Documents and to perform all Sellers' obligations arising under this Agreement and under the other Sellers Transaction Documents. Except as set forth on Schedule 6.1.2 hereto, this Agreement and the other Sellers Transaction Documents do and will each constitute legal, valid, and binding obligations of Sellers, enforceable in accordance with their respective provisions, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and by general principles of equity. Except as set forth on Schedule 6.1.2

hereto, Sellers have duly and properly taken or obtained or caused to be taken or obtained all action necessary for Sellers (a) to enter into and to deliver this Agreement, the other Sellers Transaction Documents and any and all other documents and agreements executed by Sellers in connection herewith and therewith and (b) to carry out the terms of this Agreement and the other Sellers Transaction Documents and the transaction contemplated by

- 6.1.3. No Conflict. Except as set forth on Schedules 6.1.2 and 6.1.3 hereto, Sellers' execution, delivery, and performance of this Agreement and the other Sellers Transaction Documents do not and will not (a) contravene any provision of any existing law or regulation, judgment, order, decree, or injunction, or other legal requirement, (b) result in a breach of or require consent pursuant to any lease, indenture, mortgage, agreement, guaranty, or other document by which Sellers or the Premises is bound or otherwise affected, or (c) require a filing or registration with, or the consent or approval of, any Government Entity other than with respect to any applicable state licensing agencies.
- 6.1.4. <u>Title to the Properties</u>. To Sellers' Knowledge, Sellers have not engaged in any act or omission to act so as to cause any encumbrances or restrictions to affect or relate to the Real Estate or the Properties other than the Permitted Exceptions and those mortgages, liens, encumbrances and security interests to be released or satisfied in accordance with Section 7 hereof; <u>provided</u>, <u>however</u>, that issuance of the Title Insurance Policies by the Title Company to Buyers at the Closing, as described in Section 8.1.4 hereof, shall be deemed to satisfy the obligations of Sellers with respect to title to the Properties under Section 7.1 hereof and, as such, the representation in this Section 6.1.4 shall not survive the Closing.
- 6.1.5. <u>Licenses and Accreditations</u>. To Sellers' Knowledge, based solely on the lack of receipt of written notices addressed to Sellers' Office to the contrary, there is not currently pending or threatened (a) any action or proceeding to revoke, withdraw or suspend any of the Property Licenses or (b) any judicial or administrative agency judgment or decision not to renew any of the Property Licenses. As used herein, the term "Property Licenses" shall mean all local, state and federal licenses and Medicare and Medicaid accreditations necessary to operate the Facilities.
- 6.1.6. Real Estate. Except as set forth on Schedule 6.1.6 hereto, to Sellers' Knowledge, Sellers have received no written notices that any part of the Real Estate is currently subject to condemnation proceedings and no such condemnation or taking is threatened or contemplated. To Sellers' Knowledge, Sellers have received no written notices of any public improvements which may result in special assessments against or otherwise affect the Real Estate. Other than Permitted Exceptions, the matters disclosed on the Surveys, and as set forth on Schedule 6.1.6 hereto, to Sellers' Knowledge, based solely upon the lack of receipt of written notice addressed to Sellers' Office to the contrary, there are no facts that would adversely affect the possession, use or occupancy of the Real Estate.

- 6.1.7. <u>Litigation</u>. Except as set forth on <u>Schedule 6.1.7</u> hereto, (a) Sellers are not currently (i) a party to any lawsuits, claims, administrative proceedings or other pending proceedings, or (ii) bound by any orders, judgments, injunctions, decrees or settlement agreements under which Sellers may have continuing obligations as of the date of this Agreement or as of the date of Closing, any or all of which under subsections (i) or (ii) are likely to restrict or affect in any material respect the ownership of the Properties, (b) to Sellers' Knowledge, based solely on the lack of receipt of written notices addressed to Sellers' Offices to the contrary, there (i) exists no violation by Sellers or against the Real Estate of any law, rule, regulation, ordinance or order of any court or governmental department, commission, board, bureau, agency or instrumentality, or (ii) are no lawsuits, claims, administrative proceedings or other proceedings now pending or threatened involving Sellers or the Properties, any or all of which under subsections (i) or (ii) are likely to restrict or affect in any material respect the ownership of the Properties. To Sellers' Knowledge, the right or ability of Sellers to consummate the transactions contemplated by this Agreement has not been challenged by any Governmental Entity or any other person. Notwithstanding anything contained in this Section 6.1.7, this section shall not be deemed to apply to, relate to, or involve any litigation, claim or proceeding (threatened or pending) (1) to which any of the Tenants is a party and none of Sellers is a party, or (2) that relates solely to the operations or leasing of the Facilities on or after November 1, 2002.
- 6.1.8. <u>Representations and Warranties as of the Closing</u>. The representations and warranties made by Sellers pursuant to this Section 6.1 shall be true and correct in all material respects on and as of the date of the Closing as though such representations and warranties had been made on and as of the date of the Closing.
 - 6.2. <u>Representations and Warranties by Buyers</u>. Buyers represent and warrant to Sellers as follows:
- 6.2.1. Organization. Buyers are each a limited liability company duly organized and validly existing under the laws of the state noted for each of Buyers on Exhibit B hereto.
- 6.2.2. Power and Authority. Buyers have the right, power, and authority to execute and deliver this Agreement and the other Buyers Transaction Documents and to perform all Buyers' obligations arising under this Agreement and under the Buyers Transaction Documents. This Agreement and the other Buyers Transaction Documents do and will each constitute legal, valid, and binding obligations of Buyers, enforceable in accordance with their respective provisions, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and by general principles of equity. Buyers have duly and properly taken or obtained or caused to be taken or obtained all action necessary for Buyers (a) to enter into and to deliver this Agreement, the other Buyers Transaction Documents and any and all other documents and agreements executed by Buyers in connection herewith and

therewith and (b) to carry out the terms of this Agreement and the other Buyers Transaction Documents and the transaction contemplated by them.

- 6.2.3. <u>No Conflict</u>. Buyers' execution, delivery, and performance of this Agreement and the other Buyers Transaction Documents do not and will not (a) contravene any provision of any existing law or regulation, judgment, order, decree, injunction or other legal requirement, (b) result in a breach of or require consent pursuant to any lease, indenture, mortgage, agreement, guaranty or other document by which Buyers are bound or otherwise affected, or (c) require a filing or registration with, or the consent or approval of, any Governmental Entity.
- 6.2.4. <u>Solvency</u>. To Buyers' Knowledge, Omega is not insolvent nor will Omega be rendered insolvent by the consummation of the transactions contemplated by this Agreement.
- 6.2.5. <u>Governmental Entity Challenges</u>. To Buyers' Knowledge, the right or ability of Buyers to consummate the transactions contemplated by this Agreement has not been challenged by any Governmental Entity or any other person.
- 6.2.6. Representations and Warranties as of the Closing. The representations and warranties made by Buyers pursuant to this Section 6.2 shall be true and correct in all material respects on and as of the date of the Closing as though such representations and warranties had been made on and as of the date of the Closing.

7. Title.

- 7.1. Permitted Exceptions. The Properties are sold and are to be conveyed subject to the matters listed on Schedule 7.1 hereto (collectively, the "Permitted Exceptions"). It is a condition precedent to the obligation of Buyers to consummate the transactions hereunder at the Closing that Sellers shall convey good, indefeasible and insurable fee simple title to the Real Estate to Buyers at the Closing, free and clear of all liens, mortgages and security interests and subject only to the Permitted Exceptions. The Personalty shall be conveyed to Buyers at the Closing free and clear of all liens and security interests, subject, however, to Tenants' compliance under the applicable terms of the Litchfield Master Leases.
- 7.2. <u>Standard</u>. Sellers shall give and Buyers shall accept such title as the Title Company shall approve and insure under the usual form of policy, at regular rates, subject to the usual title-policy exclusions and exceptions and the Permitted Exceptions.
 - 7.3. Evidence of Title; Surveys and Environmental Reports; Omega Board Approval.
 - 7.3.1. <u>Title Reports and Commitments</u>. After delivery of all the documents described in Section 2 hereof, Buyers may order title reports and commitments (collectively, the "Commitments"), including appropriate UCC and litigation searches of Sellers and the Properties, zoning letters, and also requesting

legible copies of all title exceptions referred to in the Commitments, from the Title Company and shall cause copies of the Commitments, UCC and litigation searches, zoning letters, and supporting documentation to be promptly forwarded to Sellers and Sellers' counsel.

- 7.3.2. <u>Title Insurance Policies</u>. The premiums for the Title Insurance Policies to be issued to Buyers at the Closing shall be paid by Sellers.
- 7.3.3. Surveys and Environmental Reports. After execution of this Agreement, Sellers shall provide Buyers with copies of all the existing Surveys and environmental reports for the Properties in Sellers' possession. After delivery of all the documents described in Section 2 hereof, Buyers shall order new updated Surveys and new environmental reports (and Buyers shall cause such environmental reports to acknowledge that Sellers are deemed to be third party beneficiaries of such environmental reports). The cost to update the Surveys shall be paid by Sellers (provided, however, in the event that the transactions contemplated by this Agreement are not consummated as a result of Buyers terminating this Agreement for reasons other than under subsection (i) of Section 10.1 hereof, then Buyers shall reimburse Sellers for the cost to update the Surveys) and the cost to update the environmental reports shall be paid by Buyers. Consistent with the terms of the Litchfield Master Leases, Sellers shall cooperate with Buyers with respect to obtaining, re-certifying or updating the Surveys and the environmental reports.
- 7.4. Title Matters and Updated Surveys. Subject to the terms of this Section 7.4 and Buyers' standard of conduct described in Section 8.3 hereof, Buyers shall have until (a) July 17, 2006 to notify Sellers of any objections that Buyers may have to matters described in those Commitments and updated Surveys received by Buyers before July 10, 2006 and (b) the Closing to notify Sellers of any objections that Buyers may have to matters described in (i) those Commitments and updated Surveys received by Buyers on or after July 10, 2006, or (ii) any new matters described in any amendments, supplements or modifications to those Commitments and updated Surveys received before July 10, 2006, and not previously disclosed to Buyers. Any title encumbrances or exceptions which are set forth in the marked up Commitments delivered by the Title Company at the Closing and in the final updated Surveys shall be included as Permitted Exceptions. In the event that, within the time period described above, Buyers object to any matters described in the Commitments and the updated Surveys, then Sellers shall have until the Closing in which to satisfy Buyers' objections; provided, however, that Sellers may elect to extend the date of Closing for up to fifteen (15) days in order to satisfy those objections from Buyers covered by subsection (b) above. In the event that Sellers shall be unable or unwilling (it being acknowledged that Sellers only obligation is to comply with the standard of conduct described in Section 8.3 hereof; provided, however, Sellers shall have no obligation to expend any monies to cure any objections) to satisfy Buyers' objections, then Buyers shall have the option to either (i) waive Buyers' objections and purchase the Properties at the Closing, in which case the waived objections shall become Permitted Exceptions or (ii) elect to terminate this Agreement. Upon election to terminate this Agreement, this Agreement shall be deemed canceled and neither party shall have any further obligations to

the other under this Agreement, except those specifically stated to survive the termination of this Agreement.

- 7.5. Removal of Mortgages, Liens and Other Encumbrances. If at the Closing there are any mortgages, liens or other encumbrances that Sellers are obligated to pay and discharge, then Sellers shall use a portion of the Purchase Price to satisfy the same and shall concurrently deliver to Buyers at the Closing instruments in recordable form sufficient to satisfy such mortgages, liens and encumbrances of record (and sufficient for the Title Company to omit the same from Buyers' Title Insurance Policies), together with the cost of recording or filing those instruments. The existence of any such mortgages, liens and other encumbrances shall not be considered objections to title if Sellers shall have complied with the foregoing requirements.
- 7.6. <u>Payment of Purchase Price</u>. Buyers shall provide all funds to be paid at the Closing pursuant to Section 5 hereof through wire transfer of immediately available funds to the Title Company.
- 7.7. Omega Board of Directors Resolutions. Within forty-five (45) days after execution of this Agreement by Sellers and Buyers, Buyers shall provide Sellers with a copy of the resolutions of the Board of Directors of Omega approving the transactions contemplated by this Agreement.

8. Conditions Precedent.

- 8.1. <u>For Buyers</u>. The obligations of Buyers under this Agreement are subject to the satisfaction on or prior to the Closing Date of the following conditions, any one or more of which may be waived in writing by Buyers to the extent permitted by applicable law:
 - 8.1.1. Representations and Warranties. The representations and warranties of Sellers set forth in this Agreement and the other Sellers Transaction Documents shall be true and correct in all material respects on and as of the date of the Closing as though made on and as of the date of the Closing, except to the extent any such representation or warranty expressly is made as of an earlier date or with respect to a particular period, in which case such representation or warranty shall have been true and correct in all material respects as of such date or with respect to such period.
 - 8.1.2. <u>Performance</u>. Sellers shall have performed or complied in all material respects with all covenants and agreements required by this Agreement and the other Sellers Transaction Documents to be performed or complied with by Sellers on or prior to the date of the Closing.
 - 8.1.3. <u>Due Diligence</u>. Buyers shall be satisfied in their sole discretion with the results of all due diligence with respect to the Properties, including the financial condition and operating history of the Tenants and the Facilities, the state of title for the Properties and the environmental and physical condition of the Properties.

- (a) Buyers shall have the period from the date this Agreement is signed by Sellers and Buyers until July 27, 2006 (the "Inspection Period"), in which to conduct due diligence with respect to the financial condition and operating history of Tenants and the Properties, the environmental and physical condition of the Properties, determine the appraised value of the Properties, and review other matters affecting Tenants and the Properties, including, but not limited to, the following: (i) obtain access to and review all books and records of Tenants and the Properties in the custody or control of Sellers and Tenants as may be reasonably required by Buyers, (ii) order new environmental reports (Phase I and, if required, Phase IIs) for the Properties, (iii) order new property condition assessments and engineering reports for the Properties, (iv) obtain copies of Tenants' current insurance policies and certificates, (v) order new MAI appraisals of the Properties, (vi) confirm the licensure, certificate of need and Medicare/Medicaid reimbursement status of Tenants and the Facilities, (vii) confirm any licensure or certificate of need change of ownership issues in Colorado, Florida, Idaho, Louisiana or Texas resulting from the sale of the Properties and the execution of the Omega Master Leases and the Peak Assignment Agreement, and (viii) such other due diligence activities materials and information as Buyers may conduct, obtain or otherwise reasonably request from Sellers and/or Tenants.
- (b) During the Inspection Period, and thereafter until the Closing hereof, so long as this Agreement has not been terminated, Sellers shall provide (or cause Tenants to provide, consistent with the terms of the Litchfield Master Leases) Buyers, their officers, employees, members, agents, contractors, engineers, consultants, licensees and assignees (i) access to all the books, contracts, and other records of Sellers (as they relate to the Tenants, the Facilities and the Properties), the Tenants, the Facilities and the Properties as Buyers may reasonably request and require in connection with the due diligence activities described in subsection (a) above and (ii) access to the Properties for the purpose of making any and all examinations and investigations of the Properties as Buyers may reasonably request and require in connection with the due diligence activities described in subsection (a) above; provided, however, that access to the Properties by Buyers must be approved, in advance, by the applicable Tenant of such Property being accessed, which approvals Sellers shall obtain for Buyers, consistent with the terms of the Litchfield Master Leases. Buyers covenant that the due diligence activities shall be undertaken in a manner as to not interfere in any material respect with the operations of the Properties.
- (c) Buyers shall not suffer or permit the filing of any mechanic's liens as a result of Buyers' due diligence activities under this Section 8.1.3 (and to the extent such mechanic's liens are or may be filed as a result of such due diligence activities, such mechanic's liens, or the right to file such liens, shall be deemed Permitted Exceptions). In the event Buyers fail to close the transactions contemplated by this Agreement, Buyers shall be liable to Sellers for the repair of any damage to the Properties occurring as a result of any such due diligence activities.

- 8.1.4. <u>Title Insurance Policies</u>. The Title Company shall have issued the Title Insurance Policies to Buyers with respect to the Properties in the form of the marked up Commitments signed and delivered at the Closing.
- 8.1.5. <u>Delivery of Sellers Closing Items</u>. Sellers shall have delivered or caused to be delivered at the Closing the items described in Section 9.2 hereof.
- 8.2. <u>For Sellers</u>. The obligations of Sellers under this Agreement are subject to the satisfaction on or prior to the Closing Date of the following conditions, any one or more of which may be waived in writing by Sellers to the extent permitted by applicable law:
 - 8.2.1. Representations and Warranties. The representations and warranties of Buyers set forth in this Agreement and the other Buyers Transaction Documents shall be true and correct in all material respects on and as of the date of the Closing as though made on and as of the date of the Closing, except to the extent any such representation or warranty expressly is made as of an earlier date or with respect to a particular period, in which case such representation or warranty shall have been true and correct in all material respects as of such date or with respect to such period.
 - 8.2.2. <u>Performance</u>. Buyers shall have performed or complied in all material respects with all covenants and agreements required by this Agreement and the other Buyers Transaction Documents to be performed or complied with by Buyers on or prior to the date of the Closing.
 - 8.2.3. <u>Delivery of Buyers Closing Items</u>. Buyers shall have delivered or caused to be delivered at the Closing the items described in Section 9.3 hereof.
- 8.3. <u>Buyers' and Sellers' Due Diligence and Commercially Reasonable Efforts in Satisfying Conditions Precedent</u>. Buyers must use due diligence and commercially reasonable efforts in seeking to satisfy all the conditions precedent set forth in Section 8.1 hereof. In addition, except as otherwise specifically set forth in Section 7.4, Sellers must use due diligence and commercially reasonable efforts in seeking to satisfy all the conditions precedent set forth in Section 8.2 hereof.

9. The Closing.

9.1. Closing and Closing Date.

9.1.1. The closing of the transaction under this Agreement (the "Closing"), shall take place on a date mutually agreed upon by Sellers and Buyers in accordance with the procedure described in Section 5.2 hereof, by delivery of the appropriate Closing documents and funds to the Title Company, but in no event shall the Closing occur later than August 1, 2006, unless such date is extended by mutual agreement of Sellers and Buyers. Upon consummation of the transactions contemplated by this Agreement, the Closing shall be deemed to be effective and the transfer of the Property shall be deemed to have occurred as of 12:01 a.m. local time on the actual Closing date agreed to by Sellers and Buyers (the "Closing Date").

- 9.2. <u>Sellers' Deliveries</u>. At the Closing, Sellers shall deliver or cause to be delivered in accordance with Section 5.2 hereof to Buyers the following, all in form and substance reasonably acceptable to Sellers and Buyers:
 - 9.2.1. The Deeds for the Real Estate, duly executed and acknowledged by Sellers and in proper form for recording, so as to convey to Buyers fee simple title to the Real Estate, free of all encumbrances other than the Permitted Exceptions;
 - 9.2.2. Mortgage discharges and UCC-3 Termination Statements in respect of all mortgages, liens or other encumbrances on the Properties in proper form for recording;
 - 9.2.3. The Bills of Sale, duly executed and acknowledged by Sellers, transferring to Buyers all of the Personalty;
 - 9.2.4. The Letter of Credit, in the amount of Three Million Dollars (\$3,000,000), for the benefit of Buyers;
 - 9.2.5. A legal opinion from counsel to Sellers regarding the matters described in Sections 6.1.1, 6.1.2 and 6.1.3 hereof;
 - 9.2.6. Such affidavits as the Title Company shall reasonably require in order to omit from its Title Insurance Policies all exceptions, except for standard exceptions, judgments, bankruptcies or other returns against persons or entities whose names are the same as or similar to Sellers' name;
 - 9.2.7. A certification of non-foreign status, in form required by the Internal Revenue Code Section 1445 and the regulations issued thereunder, duly executed by Sellers under penalty of perjury. Sellers understand that such certification will be retained by Buyers and will be made available to the Internal Revenue Service on request;
 - 9.2.8. A certificate of Sellers to the effect that the representations and warranties of Sellers set forth in this Agreement and the other Sellers Transaction Documents, are true and correct in all material respects as of the date of Closing;
 - 9.2.9. The original certificates of title to any motor vehicles included within the Personalty relating to the Properties;
 - 9.2.10. A copy of the written notice of the Peak Assignment Agreement provided by the applicable Sellers to Peak No. 2; and
 - 9.2.11. Such other documents as Buyers may reasonably request in order to further the purposes of this Agreement, duly executed and, where appropriate, acknowledged, and in recordable form.

- 9.3. <u>Buyers' Deliveries</u>. At the Closing, Buyers shall deliver to Sellers the following or cause to be delivered in accordance with Section 5.2 hereof to Sellers the following, all in form and substance reasonably acceptable to Sellers and Buyers:
 - 9.3.1. The Purchase Price;
 - 9.3.2. A certificate of Buyers to the effect that the representations and warranties of Buyers set forth in this Agreement and the other Buyers Transaction Documents, are true and correct as of the date of Closing;
 - 9.3.3. Such other documents as Sellers may reasonably request in order to further the purposes of this Agreement, duly executed and, where appropriate, acknowledged, and in recordable form; and
 - 9.3.4. A legal opinion from counsel to Buyers regarding the matters described in Sections 6.2.1, 6.2.2 and 6.2.3 hereof.
- 9.4. <u>Possession</u>. At the Closing, Sellers shall deliver to Buyers possession of the Properties, free and clear of all leases, purchase options, rights of first refusal, tenancies, and/or other rights of occupancy or acquisition of any kind (other than the Omega Master Leases, the Peak Master Lease and any resident leases), and otherwise in the condition required by this Agreement, together with all of the Personalty located at the Properties. Sellers covenant that they will not remove any Personalty from the Properties after the date of this Agreement and through the Closing Date, except items replaced or upgraded with items of equal or greater value.
 - 9.5. <u>Transfer Costs; Tax and Information Returns.</u>
 - 9.5.1. All real property transfer taxes, sales taxes, and any other similar taxes with respect to the transfer of the Properties and the Personalty shall be paid for at the Closing by Sellers.
 - 9.5.2. Sellers and Buyers shall duly execute, acknowledge where appropriate, and deliver all tax returns and information returns necessary and proper in connection with this transaction.
 - 9.5.3. The provisions of this Section 9.5 shall survive the Closing.
- 9.6. <u>Concurrent Conditions</u>. All deliveries, exchanges, and other actions to be taken at the Closing shall be deemed to occur simultaneously, and the performance of all such actions shall be concurrent conditions to all other such actions.

10. Remedies.

10.1. <u>Right to Terminate this Agreement</u>. In addition to the termination right set forth in Section 7.4 hereof, this Agreement and the transactions contemplated hereby may be terminated by written notice given in accordance with Section 14.17 hereof, as follows:

(a) b	by the mutual written agreement of Buyers and Sellers at any time prior to the Closing;
. ,	by either Buyers or Sellers if the requirements in the first sentence of Section 2 hereof are not satisfied on or if the requirements in the second sentence of Section 2 hereof are not satisfied on or before May 15, 2006;
` ,	by Buyers if the requirements under Section 3 hereof are not satisfied on or before May 25, 2006; provided, terminate this Agreement under this subsection (c) as a result of Peak No. 2 exercising its rights to acquire

- (d) by Buyers if the requirements under Section 4 hereof are not satisfied;
- (e) by either Sellers or Buyers if the Closing shall not have occurred at or before 11:59 p.m. EDT on August 1, 2006, unless such date is extended by mutual agreement of Sellers and Buyers;
 - (f) by Buyers if a Material Adverse Change occurs or is found to have occurred after the Relevant Date;
- (g) by Buyers or Sellers if the Board of Directors of Omega fails to approve the transactions contemplated by this Agreement and in the timeframe set forth in Section 7.7 hereof;
- (h) by Buyers prior to 11:59 p.m. EDT on July 27, 2006 as a result of the examinations and investigations of Tenants and the Properties described in Section 8.1.3(a) hereof and conducted during the Inspection Period;
- (i) by Buyers, by reason of the breach, inaccuracy or non-fulfillment of any representation, warranty, covenant, obligation or agreement by Sellers under this Agreement; or
- (j) by Sellers, by reason of the breach, inaccuracy or non-fulfillment of any representation, warranty, covenant, obligation or agreement by Buyers under this Agreement.
- 10.2. <u>Liability for Agreement Termination</u>. If the parties terminate this Agreement pursuant to subsections (a) through (h) of Section 10.1 hereof, then neither party shall have any further obligations to the other under this Agreement, except those specifically stated to survive the termination of this Agreement. Notwithstanding the foregoing, in the event that this Agreement is terminated under subsections (i) or (j) of Section 10.1 hereof, then the terminating party, if successful in proving its right to terminate under the applicable subsection, may recover its actual damages (but in no event consequential (e.g., profits alleged to have been realized had the transactions contemplated

herein been consummated according to their terms), incidental, special, punitive or exemplary damages) incurred as a result of such termination.

11. Closing Date Prorations.

- 11.1. <u>Items to be Prorated at Closing</u>. If the Closing Date occurs on any date other than the first day of a calendar month, then all Rent as defined in and with respect to the Peak Master Lease shall be prorated between the applicable Sellers and OHI-CO as of the Closing Date, with appropriate credits or charges made at the Closing in accordance with Section 5.2.4 hereof.
- 11.2. <u>Errors.</u> Any errors in calculations, apportionments or payments of Rent as defined in and with respect to the Peak Master Lease shall be corrected, adjusted and paid as soon as practicable after discovery, whether before or after the Closing Date. This Section 11.2 shall survive the Closing and any payments made by Sellers or OHI-CO under this Section 11 shall not be included in calculating the limitations on damages under Sections 16.3 or 16.7 hereof, as applicable.
- 12. <u>Further Assurances</u>. The parties shall do such other and further acts and things, and to execute and deliver such instruments and documents, as either may reasonably request from time to time, whether at or after the Closing, in furtherance of the purposes of this Agreement and the other Transaction Documents. The obligation under this Section 12 shall survive the Closing.
- 13. <u>Brokers</u>. Sellers and Buyers represent and warrant to each other that each has dealt with no broker or finder in connection with this Agreement or any of the transactions contemplated hereby, and knows of no broker or finder entitled to or claiming a fee, commission, or other similar compensation in connection with this Agreement or any of the transactions contemplated hereby. Sellers and Buyers shall indemnify and hold harmless each other and any affiliates or representatives of the other from and against liabilities, damages, and costs (including reasonable attorneys' fees) arising out of any claim for commissions or other compensation made by any broker or finder who shall claim to have dealt with the indemnifying party in connection with this Agreement or any of the transactions contemplated hereby. The provisions of this Section 13 shall survive the Closing or any termination of this Agreement.

14. Miscellaneous.

- 14.1. <u>Complete Agreement; No Other Representations, Warranties or Covenants</u>. This Agreement and the other Transaction Documents constitute the entire agreement between the parties with respect to the transactions contemplated hereunder and shall supersede all prior oral or written agreements and all contemporaneous oral negotiations, commitments and understandings. Neither Buyers nor Sellers shall be deemed to have made to the other any representations or warranties that are not expressly set forth in this Agreement or the other Transaction Documents.
- 14.2. <u>No Reliance</u>. Each party hereto expressly disclaims reliance upon any facts, promises, undertakings or representations made by any other party, or its agents, representatives or attorneys prior to the execution of this Agreement.

- 14.3. <u>Amendment</u>. This Agreement may not be amended, waived, superseded, renewed, extended or terminated orally, but only by an agreement in writing signed by each party or, in the case of a waiver, by the party waiving compliance.
- 14.4. <u>Captions</u>. The captions or headings contained in this Agreement are for convenience and reference only and shall not affect the interpretation of this Agreement.
- 14.5. <u>Governing Law: Jurisdiction</u>. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HERETO SHALL BE GOVERNED, CONSTRUED AND ENFORCED IN ACCORDANCE WITH THE DOMESTIC SUBSTANTIVE LAWS OF THE STATE OF MARYLAND, WITHOUT GIVING EFFECT TO ANY CHOICE OR CONFLICTS OF LAW PROVISION OR RULE THAT WOULD CAUSE THE APPLICATION OF THE DOMESTIC SUBSTANTIVE LAWS OF ANY OTHER JURISDICTION. SELLERS AND BUYERS HEREBY ABSOLUTELY AND IRREVOCABLY CONSENT AND SUBMIT TO THE JURISDICTION OF THE STATE COURTS OF BALTIMORE COUNTY, MARYLAND AND/OR, IF APPLICABLE, ANY FEDERAL COURT LOCATED IN SUCH COUNTY IN THE STATE OF MARYLAND, IN CONNECTION WITH ANY ACTIONS OR PROCEEDINGS ARISING OUT OF OR RELATING TO THIS AGREEMENT. IN SUCH ACTION OR PROCEEDING, SELLERS AND BUYERS HEREBY ABSOLUTELY AND IRREVOCABLY WAIVE ANY OBJECTION TO VENUE.

14.6. Certain Definitions.

- 14.6.1. The words "hereof," "herein," and "hereunder," and words of similar import, shall be construed to refer to this Agreement as a whole, and not to any particular Section or provision, unless otherwise expressly provided.
- 14.6.2. The word "person" when used in this Agreement shall mean any natural person, partnership, limited liability company, trust, corporation, or other form of business or legal entity.
 - 14.6.3. The word "mortgage" shall be deemed to mean either a mortgage or deed of trust.
 - 14.6.4. The word "including" shall be construed as being followed by the words "but not limited to" or "without limitation."
- 14.7. <u>Number/Gender</u>. All words or terms used in this Agreement, regardless of the number or gender in which they are used, shall be deemed to include any other number and any other gender as the context may require.
- 14.8. <u>Successors and Assigns</u>. This Agreement shall be binding upon and shall inure to the benefit of Sellers and Buyers and their respective successors and permitted assigns. Neither this Agreement, nor any rights, interests or obligations hereunder, may be assigned or transferred, in whole or in part, by operation of law or otherwise by Buyers or Sellers without the prior written consent of the other party and any such assignment that is not consented to shall be null and void.

- 14.9. <u>Counterparts</u>. This Agreement may be executed in multiple counterparts, each of which when so executed and delivered shall be an original, but all of which shall evidence a single agreement.
- 14.10. <u>Exhibits and Schedules</u>. All Exhibits and Schedules referred to in this Agreement are incorporated into this Agreement by reference and shall be deemed part of this Agreement for all purposes as if set forth at length in this Agreement.
- 14.11. <u>No Joint Venture, Partnership, Agency, Etc.</u> This Agreement shall not be construed as in any way establishing a partnership, joint venture, express or implied agency, or employer-employee relationship between Sellers and Buyers.
- 14.12. <u>No Third-Party Beneficiaries</u>. This Agreement is for the sole benefit of Sellers and Buyers and their respective successors and permitted assigns, and no other person or entity shall be entitled to rely upon, enforce, or receive any legal or equitable right, remedy or claim under or in respect of this Agreement or any provision hereof.
- 14.13. <u>No Waiver</u>. No delay on the part of any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof, nor shall any waiver on the part of any party of any right, power or privilege, nor any single or partial exercise of any such right, power or privilege, preclude any further exercise thereof or the exercise of any other such right, power or privilege.
- 14.14. <u>Remedies</u>. Subject to the terms and conditions of Section 16 hereof, all of the rights and remedies of either party under this Agreement and the other Transaction Documents are intended to be distinct, separate, and cumulative, and no such right or remedy herein or therein mentioned is intended to be in exclusion of or a waiver of any other rights or remedies that either party may otherwise have at law or in equity.
- 14.15. <u>No Presumption</u>. This Agreement is the result of extensive negotiations between the parties, which were all represented by counsel. The parties acknowledge and agree that in the event that any dispute arises regarding the interpretation or construction of this Agreement, this Agreement shall not be strictly construed against any party hereto by reason of the rule of construction that a document is to be construed more strictly against the party which drafted the agreement.
- 14.16. <u>Invalidity</u>. If any term, provision, covenant or restriction of this Agreement or the application thereof to any person or circumstance shall, to any extent, be held by a court of competent jurisdiction to be invalid, void or unenforceable, Sellers and Buyers shall direct that such court interpret and apply the remainder of this Agreement in the manner that it determines most closely effectuates their intent in entering into this Agreement, and in doing so particularly take into account the relative importance of the term, provision, covenant or restriction being held invalid, void or unenforceable.
- 14.17. Notices. All notices, requests, demands, waivers and other communications required or to be permitted to be given under this Agreement shall be in writing and shall be deemed to have been duly given: (a) if delivered personally or sent by facsimile or e-mail, on the date

received, (b) if delivered by overnight courier, on the date received, and (c) if mailed, three (3) days after mailing with postage prepaid. Any such notice shall be sent as follows:

If to Sellers: Litchfield Investment Company, L.L.C.

24 Bank Street

New Milford, Connecticut 06776 Attention: Bruce Weinstein Telephone: 800-350-0725

E-mail: bruce@litchfieldcompanies.com

With a copy to: Owens, Clary & Aiken, L.L.P.

700 North Pearl Street, Suite 1600

Dallas, Texas 75201

Attention: Leighton Aiken, Esq. Telephone: 214-698-2103 E-mail: laiken@oca-law.com

If to Buyers: c/o Omega Healthcare Investors, Inc.

9690 Deereco Road, Suite 100 Timonium, Maryland 21093 Attention: Daniel J. Booth Telephone: 410-427-8824

E-mail: dbooth@omegahealthcare.com

With a copy to: LeBoeuf, Lamb, Greene & MacRae LLP 125 West 55th Street

New York, New York 10019
Attention: John R. Fallon, Jr., Esq.
Telephone: 212-424-8279
E-mail: jrfallon@llgm.com

or to such other person and/or address as shall have been specified by either party in a notice given to the other; <u>provided</u>, that such notice shall be effective only upon receipt.

- 14.18. <u>No Recordation of this Agreement</u>. Neither Buyers nor Sellers shall record this Agreement or a memorandum of this Agreement.
- 14.19. <u>Time Periods</u>. If the time period by which any acts or payments required hereunder must be performed or paid expires on a Saturday, Sunday or legal holiday, then such time period shall be automatically extended to the close of business on the next regularly scheduled business day.
- 14.20. <u>Legal and Transaction Costs</u>. Without regard to whether (a) the transaction closes or (b) if the transaction does not close, the reason why it does not close, the responsibility for transaction costs and expenses, as between Sellers and Buyers shall be as follows: Sellers and Buyers shall pay for their own legal counsel fees and expenses, Sellers shall pay the premium costs for the title insurance to be issued to Buyers and the costs and expenses to update the Surveys, and Buyers shall pay all other due diligence

costs, including appraisals, UCC and litigation searches, zoning letters, and environmental and property condition reports. All real property transfer taxes, sales taxes and other taxes regarding transfer of the Properties shall be paid by Sellers and, in accordance with Section 11.1 hereof, all Rent as defined in and with respect to the Peak Master Lease shall be prorated between the applicable Sellers and OHI-CO as of the Closing Date. The provisions of this Section 14.20 shall survive any termination of this Agreement or the Closing.

- 14.21. <u>Prevailing Party.</u> In the event of a dispute between Sellers and Buyers with respect to the interpretation, enforcement, construction or operation of the terms and conditions of this Agreement and the other Transaction Documents, the prevailing party in connection with the resolution of such dispute, whether in a court proceeding, arbitration or otherwise, shall be entitled to collect from the other party its reasonable attorneys' fees, costs and expenses, including its costs, fees and expenses in connection with any appeals or enforcement of an attorney's fee award.
- 14.22. Confidentiality. From the date of this Agreement, Sellers and Buyers agree to keep the contents of this Agreement and any discussions between the parties relating to this matter confidential and not to disclose the contents of this Agreement or such discussions to any third-party (except Sellers' lender, the Tenants, the attorneys, accountants or consultants hired by Sellers and Buyers and except for required SEC or other governmental agency disclosures) without the prior written consent of Sellers or Buyers, as applicable, except as may be required to enforce a party's rights hereunder or as a party may be legally required. Either Buyers' or Sellers' breach of the provisions of this Section 14.22 shall entitle the other to seek judicial or equitable remedies, including, but not limited to, injunctive relief.
- 14.23. <u>Exclusivity</u>. Sellers, on behalf of themselves and their respective affiliates, agree not to enter into discussions or negotiations with, or furnish any information to, any party regarding a transaction relating to the sale, financing, managing or leasing of the Properties, unless this Agreement is terminated.

15. <u>Sellers' and Buyers' Indemnifications</u>.

15.1. <u>Sellers' Indemnification</u>. Subject to the limitations on liability and damages set forth in Section 16 hereof, Sellers shall jointly and severally indemnify and hold harmless Buyers and their respective affiliates, officers, members, directors, shareholders, employees, agents and assigns (collectively, the "Buyers Indemnified Parties") from and against any and all damages, losses, liabilities, obligations, claims, actions, suits, proceedings, investigations, demands, assessments, judgments, penalties, sanctions, costs, expenses, and disbursements (including, without limitation, reasonable attorneys' and consultants' fees and expenses), whether or not subject to litigation (collectively, "Claims") of any kind or character imposed upon, arising out of, in connection with, incurred or in any way attributed to or relating to (a) the breach or failure of any representation, warranty, covenant or obligation of Sellers that is contained in this Agreement or any other Sellers Transaction Documents, (b) the use, operation, ownership or management of the Properties covered by the HQM Master Lease and the Nexion

Master Lease accruing or arising prior to November 1, 2002. (c) except for any matters described on Schedule 15.1(c) hereto, provided Sellers have Sellers' Knowledge regarding the following having accrued or arose prior to November 1, 2002, (i) any of the Properties not being in compliance in all material respects with any applicable Environmental Laws, (ii) Hazardous Substances being located on or at any of the Properties, (iii) Environmental Claims being made against any of the Properties, (iv) Remedial Action being taken in respect of any of the Properties and (v) any written notice or written complaint from any Governmental Entity or third-party alleging the failure of any of the Properties to comply with any Environmental Laws, (d) except for any matters described on Schedule 15.1(c) hereto, provided Sellers have Sellers' Knowledge obtained solely by virtue of written notice received by Sellers at Sellers' Office regarding the following having accrued or arose after November 1, 2002, (i) any of the Properties not being in compliance in all material respects with any applicable Environmental Laws, (ii) Hazardous Substances being located on or at any of the Properties, (iii) Environmental Claims being made against any of the Properties, (iv) Remedial Action being taken in respect of any of the Properties and (v) any written notice or written complaint from any Governmental Entity or third-party alleging the failure of any of the Properties to comply with any Environmental Laws, (e) any and all Claims asserted by The ARBA Group (including any affiliates or owners of The ARBA Group), (f) any indemnification obligations of the applicable Sellers as Landlord under Section 15.10 of the Peak Master Lease accruing or arising prior to the date of Closing, and (g) Claims asserted by Peak No. 2 with respect to the renewal provisions in Section 2.2 of the Peak Master Lease; provided, however, that Sellers indemnification obligations with respect to this subsection (g) shall be limited to payment of the first Two Hundred Fifty Thousand Dollars (\$250,000) of litigation costs and expenses incurred by OHI-CO in the defense of any lawsuit filed against OHI-CO by Peak No. 2 arising from any renewal obligations that OHI-CO may have as Landlord under Section 2.2 of the Peak Master Lease; provided, further, that any indemnification payments by Sellers under this subsection (g) shall be excluded from Sellers' limitation for damages under Section 16.3 hereof.

15.2. <u>Sellers' Indemnification Procedure.</u> If the Buyers Indemnified Parties assert that Sellers are subject to a Claim for indemnification pursuant to Section 15.1, the Buyers Indemnified Parties shall promptly notify Sellers in writing of the Claim and shall describe in such notice the Claim in sufficient detail in order to permit Sellers to evaluate the nature and cause of the Claim. Sellers covenant and agree to defend, through counsel retained by Sellers, which retention shall be reasonably approved by Buyers, the Buyers Indemnified Parties on account of any of said Claims and to pay any judgment against the Buyers Indemnified Parties, or any other amount that is indicated in Section 15.1, along with all reasonable costs and expenses relative to any Claims, including attorneys' fees and expenses; <u>provided</u>, that the Buyers Indemnified Parties shall, nevertheless, have the right, if they so elect and at their sole cost and expense, to participate (with counsel of their choosing, which counsel must be approved by Sellers, which approval may not be unreasonably withheld) in the defense of any such Claims in which they may be a party without relieving Sellers of the obligation to defend same. To the extent applicable, the Buyers Indemnified Parties covenant not to settle or compromise any Claims without the prior written consent of Sellers, which consent may not be unreasonably withheld. Failure to comply with the preceding covenant shall be deemed a complete waiver of any rights

that the Buyers Indemnified Parties have or may have under Sections 15.1 and 15.2. Any required indemnification payment shall be made within fifteen (15) days after liability for and the amount of the indemnification payment is finally determined.

- 15.3. <u>Buyers' Indemnification</u>. Subject to the limitations on liability and damages set forth in Section 16 hereof, Buyers shall jointly and severally indemnify and hold harmless Sellers and their respective affiliates, officers, members, directors, shareholders, employees, agents and assigns (collectively, the "Sellers Indemnified Parties") from and against any and all Claims of any kind or character imposed upon, arising out of, in connection with, incurred or in any way attributed to or relating to (a) the breach or failure of any representation, warranty, covenant or obligation of Buyers that is contained in this Agreement or any other Buyers Transaction Documents, (b) any activities conducted at or upon the Properties involving due diligence activities by or for Buyers under this Agreement, except if such Claims are caused by the gross negligence or willful misconduct of Tenants and (c) any indemnification obligations of OHI-CO as Landlord under Section 15.10 of the Peak Master Lease accruing or arising on or after the date of Closing.
- 15.4. <u>Buyers' Indemnification Procedure.</u> If the Sellers Indemnified Parties assert that Buyers are subject to a Claim for indemnification pursuant to Section 15.3, the Sellers Indemnified Parties shall promptly notify Buyers in writing of the Claim and shall describe in such notice the Claim in sufficient detail in order to permit Buyers to evaluate the nature and cause of the Claim. Buyers covenant and agree to defend, through counsel retained by Buyers which retention shall be reasonably approved by Sellers, the Sellers Indemnified Parties on account of any of said Claims and to pay any judgment against the Sellers Indemnified Parties, or any other amount that is indicated in Section 15.3, along with all reasonable costs and expenses relative to any Claims, including attorneys' fees and expenses; <u>provided</u>, that the Sellers Indemnified Parties shall, nevertheless, have the right, if they so elect and at their sole cost and expense, to participate (with counsel of their choosing, which counsel must be approved by Buyers, which approval may not be unreasonably withheld) in the defense of any such Claims in which they may be a party without relieving Buyers of the obligation to defend same. To the extent applicable, the Sellers Indemnified Parties covenant not to settle or compromise any Claims without the prior written consent of Buyers, which consent may not be unreasonably withheld. Failure to comply with the preceding covenant shall be deemed a complete waiver of any rights that the Sellers Indemnified Parties have or may have under Sections 15.3 and 15.4. Any required indemnification payment shall be made within fifteen (15) days after liability for and the amount of the indemnification payment is finally determined.

16. <u>Limitations on Liability and Payment Obligations</u>.

16.1. <u>Sellers' Limitation on Liability</u>. Under no circumstances whatsoever shall Sellers be liable for any loss of profits, loss of revenue, loss of business or other incidental, consequential, indirect, special, exemplary or punitive damages of any kind, arising out of or in connection with any and all Claims under Section 15.1 hereof, any and all other liabilities under this Agreement and the other Transaction Documents, and/or any of the transactions contemplated by this Agreement and the other Transaction Documents.

- 16.2. No Personal Liability of Sellers' Members. All representations, warranties, covenants, duties, obligations and liabilities of Sellers shall be the sole responsibility of Sellers and shall be recourse solely to Sellers and their assets; <u>provided</u>, that Buyers' sole recourse is the Letter of Credit with respect to those Claims under Section 15.1 hereof and those breaches under this Agreement and the other Transaction Documents for which Sellers' liability for damages is limited under Section 16.3 hereof. Under no circumstances whatsoever shall any member, manager, officer, director, shareholder, employee or representative of Sellers be deemed personally liable, in contract, tort or otherwise, for any such representations, warranties, covenants, duties, obligations and liabilities arising out of or under this Agreement or the other Transaction Documents.
- 16.3. <u>Limitation on Sellers' Liability for Damages.</u> EXCEPT FOR ANY BREACH OF THE REPRESENTATIONS AND WARRANTIES SET FORTH IN SECTIONS 6.1.1, 6.1.2, 6.1.3 AND 6.1.7 HEREOF AND CLAIMS UNDER SECTIONS 15.1(b), 15.1(c), 15.1(d) AND 15.1(e) HEREOF, SELLERS' MAXIMUM AGGREGATE LIABILITY FOR DAMAGES WITH RESPECT TO CLAIMS UNDER SECTION 15.1 HEREOF AND ANY BREACH OF A REPRESENTATION, WARRANTY, COVENANT, OBLIGATION, TERM OR CONDITION UNDER THIS AGREEMENT AND THE OTHER TRANSACTION DOCUMENTS IS EXPRESSLY LIMITED TO THE SUM OF THREE MILLION DOLLARS (\$3,000,000) IN THE EVENT THE CLOSING TAKES PLACE. EXCEPT FOR ANY BREACH OF THE RERESENTATIONS AND WARRANTIES SET FORTH IN SECTIONS 6.1.1, 6.1.2, 6.1.3 AND 6.1.7 HEREOF AND CLAIMS UNDER SECTIONS 15.1(b), 15.1(c), 15.1(d) AND 15.1(e) HEREOF, IT IS HEREBY AGREED TO BY SELLERS AND BUYERS THAT THIS SECTION 16.3 IS, AND SHALL SERVE AS, A LIMITATION ON SELLERS' LIABILITY FOR DAMAGES IN RESPECT OF CLAIMS UNDER SECTION 15.1 HEREOF AND ANY OTHER BREACHES DESCRIBED IN THIS SECTION 16.3.
- 16.3.1. <u>Buyers' Recourse Against Letter of Credit</u>. As set forth in this Section 16.3.1 and in the L/C Agreement, it is hereby agreed to by the parties that the Letter of Credit shall be the sole recourse for any obligation or liability owed by Sellers (or any one of them) to Buyers (or any one of them) with respect to those Claims under Section 15.1 hereof and those breaches of any representation, warranty, covenant, obligation, term or condition under this Agreement and the other Transaction Documents for which Sellers' liability for damages is limited under Section 16.3 hereof, and no other assets of Sellers (and Sellers' members) can be utilized or executed upon to satisfy or pay any liability or obligation owed by Sellers to Buyers with respect to those Claims under Section 15.1 hereof and those breaches under this Agreement and the other Transaction Documents for which Sellers' liability for damages is limited under Section 16.3 hereof.
 - 16.4. Intentionally Deleted.
 - 16.5. <u>Buyers' Limitation on Liability</u>. Under no circumstances whatsoever shall Buyers be liable for any loss of profits, loss of revenue, loss of business, or other incidental, consequential, indirect, special, exemplary or punitive damages of any kind, arising out of or in connection with all Claims under Section 15.3 hereof, all other

liabilities under this Agreement and the other Transaction Documents, and/or any of the transactions contemplated by this Agreement and the other Transaction Documents.

- 16.6. <u>No Personal Liability of Buyers' Officers</u>. All representations, warranties, covenants, duties, obligations and liabilities of Buyers shall be the sole responsibility of Buyers and shall be recourse solely to Buyers and their assets. Under no circumstances whatsoever shall any member, manager, officer, director, shareholder, employee or representative of Buyers be deemed personally liable, in contract, tort or otherwise, for any such representations, warranties, covenants, duties, obligations and liabilities arising out of or under this Agreement or the other Transaction Documents.
- 16.7. <u>Limitation on Buyers' Liability for Damages</u>. EXCEPT AS OTHERWISE PROVIDED IN SECTION 16.8 HEREOF, BUYERS' MAXIMUM AGGREGATE LIABILITY FOR DAMAGES WITH RESPECT TO CLAIMS UNDER SECTION 15.3 HEREOF AND ANY BREACH OF A REPRESENTATION, WARRANTY, COVENANT, OBLIGATION, TERM OR CONDITION UNDER THIS AGREEMENT AND THE OTHER TRANSACTION DOCUMENTS IS EXPRESSLY LIMITED TO THE SUM OF THREE MILLION DOLLARS (\$3,000,000) IN THE EVENT THE CLOSING TAKES PLACE. EXCEPT AS OTHERWISE PROVIDED IN SECTION 16.8 HEREOF, IT IS HEREBY AGREED TO BY SELLERS AND BUYERS THAT THIS SECTION 16.7 IS, AND SHALL SERVE AS, A LIMITATION ON BUYERS' LIABILITY FOR DAMAGES IN RESPECT OF CLAIMS UNDER SECTION 15.3 HEREOF AND ANY OTHER BREACHES DESCRIBED IN THIS SECTION 16.7.
- 16.8. <u>Exceptions to Limitation on Buyers' Liability for Damages</u>. BUYERS' MAXIMUM AGGREGATE LIABILITY FOR DAMAGES WITH RESPECT TO ANY BREACH OF THE REPRESENTATIONS AND WARRANTIES SET FORTH IN SECTIONS 6.2.1, 6.2.2, 6.2.3 AND 6.2.4 HEREOF IS EXPRESSLY LIMITED TO THE SUM OF THE PURCHASE PRICE; <u>PROVIDED</u>, <u>HOWEVER</u>, THAT IN ANY SUCH ACTIONS INITIATED BY SELLERS AGAINST BUYERS, SELLERS WILL NOT OBJECT TO BUYERS' INSTITUTION OF A THIRD-PARTY ACTION.
- 17. <u>Survival of Indemnifications, Obligations, Representations and Warranties</u>. Except as otherwise provided in Section 6.1.4 hereof, the indemnifications, obligations, representations and warranties contained in this Agreement and the other Transaction Documents shall survive the Closing and remain operative and in full force for a period of two (2) years from the date of Closing (the "Survival Termination Date"); <u>provided, however</u>, that any Claim, demand, lawsuit or other action must be made or asserted in writing to Sellers or Buyers, as the case may be, prior to the Survival Termination Date for such indemnifications and other liabilities and obligations of Sellers and Buyers, as the case may be, to survive the Survival Termination Date. Then, in such event, such Claim, demand, lawsuit or other action shall survive the Survival Termination Date until finally resolved.

SIGNATURE PAGE FOLLOWS

IN WITNESS WHEREOF, the parties have duly executed and delivered this Contract of Sale as of the date first above written.

SELLERS:

LARAMIE ASSOCIATES, LLC
CASPER ASSOCIATES, LLC
NORTH 12TH STREET ASSOCIATES, LLC
NORTH UNION BOULEVARD ASSOCIATES, LLC
JONES AVENUE ASSOCIATES, LLC
LITCHFIELD INVESTMENT COMPANY, L.L.C.
USTICK ROAD ASSOCIATES, LLC
WEST 24TH STREET ASSOCIATES, LLC
NORTH THIRD STREET ASSOCIATES, LLC
MIDWESTERN PARKWAY ASSOCIATES, LLC
NORTH FRANCIS STREET ASSOCIATES, LLC
WEST NASH STREET ASSOCIATES, LLC

By: /s/ Bruce Weinstein

Name: Bruce Weinstein

Title: Vice President and Member

BUYERS:

OHI ASSET (LA), LLC NRS VENTURES, L.L.C. OHI ASSET (CO), LLC

By: Omega Healthcare Investors, Inc., as the Sole Member of each of the companies

By: /s/ Daniel J. Booth

Name: Daniel J. Booth

Title: Chief Operating Officer

SCHEDULES AND EXHIBITS

Exhibit A Sellers Exhibit B Buyers

Exhibit C Descriptions of Facilities
Exhibit D Legal Descriptions of Land
Schedule 6.1.2 Power and Authority

Schedule 6.1.3 No Conflict
Schedule 6.1.6 Real Estate
Schedule 6.1.7 Litigation

Schedule 7.1 Permitted Exceptions
Schedule 15.1(c) Environmental Matters

EXHIBIT A

SELLERS

SELLERS	STATE OF ORGANIZATION		
Laramie Associates, LLC	Colorado		
Casper Associates, LLC	Colorado		
North 12 th Street Associates, LLC	Colorado		
North Union Boulevard Associates, LLC	Colorado		
Jones Avenue Associates, LLC	Colorado		
Litchfield Investment Company, L.L.C.	Connecticut		
Ustick Road Associates, LLC	Idaho		
West 24 th Street Associates, LLC	Texas		
North Third Street Associates, LLC	Texas		
Midwestern Parkway Associates, LLC	Texas		
North Francis Street Associates, LLC	Texas		
West Nash Street Associates, LLC	Texas		

EXHIBIT B

BUYERS

BUYERS	STATE OF ORGANIZATION
OHI Asset (LA), LLC	Delaware
NRS Ventures, L.L.C.	Kentucky
OHI Asset (CO), LLC	Delaware

EXHIBIT C

DESCRIPTIONS OF PROPERTIES/FACILITIES

PROPERTIES/ FACILITIES	SELLERS	TENANTS	BUYERS	PROPERTY/FACILITY ADDRESSES	PURCHASI PRICE
Cheyenne Mountain Care Center	Laramie Associates, LLC	Peak Medical Colorado No. 2, Inc.	OHI Asset (CO), LLC	835 Tenderfoot Hill Road Colorado Springs, Colorado 80906 719-576- 8380 719-576-5691 (fax)	\$9,622,000
Cheyenne Place Retirement Center	Casper Associates, LLC	Peak Medical Colorado No. 2, Inc.	OHI Asset (CO), LLC	945 Tenderfoot Hill Road Colorado Springs, Colorado 80906 719-576-2122 719-576-1352 (fax)	\$5,880,000
Mesa Manor Care Center	North 12 th Street Associates, LLC	Peak Medical Colorado No. 2, Inc.	OHI Asset (CO), LLC	2901 North 12th Street Grand Junction, Colorado 81506 970-243-7211 970-245-5104 (fax)	\$5,239,000
Pikes Peak Care Center	North Union Boulevard Associates, LLC	Peak Medical Colorado No. 2, Inc.	OHI Asset (CO), LLC	2719 North Union Boulevard Colorado Springs, Colorado 80906 719-636-1676 719-636-9168 (fax)	\$11,225,000
Pueblo Extended Care Center	Jones Avenue Associates, LLC	Peak Medical Colorado No. 2, Inc.	OHI Asset (CO), LLC	2611 Jones Avenue Pueblo, Colorado 81004 719-564-1735 719-564-1501 (fax)	\$9,034,000
Fort Myers Care & Rehabilitation Center	Litchfield Investment Company, L.L.C.	HQM of Ft. Myers, LLC	NRS Ventures, L.L.C.	13755 Gulf Club Parkway Fort Myers, Florida 33919 941-482-2848 941-482-1370 (fax)	\$7,021,000
Heritage Park Care & Rehabilitation Center	Litchfield Investment Company, L.L.C.	HQM of Bradenton, LLC	NRS Ventures, L.L.C.	2302 59th Street West Bradenton, Florida 34209 941-792-8480 941-794-8905 (fax)	\$7,874,000
HQM of Orange Park	Litchfield Investment Company, L.L.C.	HQM of Orange Park, LLC	NRS Ventures, L.L.C.	2029 Professional Center Drive Orange Park, Florida 32073 904-272-6194 904-272-2085 (fax)	\$6,889,000
HQM of Palm Bay	Litchfield Investment Company, L.L.C.	HQM of Palm Bay, LLC	NRS Ventures, L.L.C.	1515 Port Malabar Boulevard Palm Bay, Florida 32905 407-723-1235 407-951-2630 (fax)	\$7,874,000
HQM of Port Charlotte	Litchfield Investment Company, L.L.C.	HQM of Port Charlotte, LLC	NRS Ventures, L.L.C.	4033 Beaver Lane Port Charlotte, Florida 33952 941-625-3200 941-624-2358 (fax)	\$10,760,000
Kenilworth Care & Rehabilitation Center	Litchfield Investment Company, L.L.C.	HQM of Sebring, LLC	NRS Ventures, L.L.C.	3011 Kenilworth Boulevard Sebring, Florida 33870 863-382-2153 863-382-2039 (fax)	\$6,824,000

Winter Park Care & Rehabilitation Center	Litchfield Investment Company, L.L.C.	HQM of Winter Park, LLC	NRS Ventures, L.L.C.	2970 Scarlet Road Winter Park, Florida 32792 407-671-8030 407-671-3746 (fax)	\$6,758,000
Capitol Care Center	Ustick Road Associates, LLC	Peak Medical Colorado No. 2, Inc.	OHI Asset (CO), LLC	8211 Ustick Road Boise, Idaho 83704 208-375-3700 208-322-0390 (fax)	\$5,000,000
Gonzales Healthcare Center	Litchfield Investment Company, L.L.C.	Nexion Health at Gonzales, Inc.	OHI Asset (LA), LLC	905 W. Cornerview Road Gonzales, Louisiana 70737 504-647-7841 504-644-8409 (fax)	\$3,720,000
Kaplan Healthcare Center	Litchfield Investment Company, L.L.C.	Nexion Health at Kaplan, Inc.	OHI Asset (LA), LLC	1300 West 8th Street Kaplan, Louisiana 70548 318-643-2300 318-643-1579 (fax)	\$3,600,000
Lafayette Healthcare Center	Litchfield Investment Company, L.L.C.	Nexion Health at Lafayette, Inc.	OHI Asset (LA), LLC	325 Bacque Crescent Drive Lafayette, Louisiana 70503 318-232-0299 318-237-8162 (fax)	\$1,800,000
Many Healthcare North	Litchfield Investment Company, L.L.C.	Nexion Health at Many North, Inc.	OHI Asset (LA), LLC	120 Natchitoches Highway 6 East Many, Louisiana 71449 318-256-9233 318-256-0739 (fax)	\$3,840,000
Many Healthcare South	Litchfield Investment Company, L.L.C.	Nexion Health at Many South, Inc.	OHI Asset (LA), LLC	255 Middle Creek Road Many, Louisiana 71449 318-256-6281 318-256-0741 (fax)	\$1,800,000
Marrero Healthcare Center	Litchfield Investment Company, L.L.C.	Nexion Health at Marrero, Inc.	OHI Asset (LA), LLC	5301 August Lane Marrero, Louisiana 70072 504-341-3658 504-347-3754 (fax)	\$4,020,000
Meadowview Healthcare Center	Litchfield Investment Company, L.L.C.	Nexion Health at Minden, Inc.	OHI Asset (LA), LLC	400 Meadowview Drive Minden, Louisiana 71055 318-377-1011 318-377-9814 (fax)	\$6,900,000
New Iberia Manor North	Litchfield Investment Company, L.L.C.	Nexion Health at New Iberia North, Inc.	OHI Asset (LA), LLC	1803 Jane Street New Iberia, Louisiana 70560 318-365-2466 318-365-2460 (fax)	\$3,630,000
New Iberia Manor South	Litchfield Investment Company, L.L.C.	Nexion Health at New Iberia South, Inc.	OHI Asset (LA), LLC	600 Bayard Street New Iberia, Louisiana 70562 318-365-3441 318-365-0879 (fax)	\$2,400,000
Claiborne Healthcare Center	Litchfield Investment Company, L.L.C.	Nexion Health at Claiborne, Inc.	OHI Asset (LA), LLC	1536 Claiborne Avenue Shreveport, Louisiana 71101 318-631-3426 318-636-4936 (fax)	\$2,580,000
Thibodaux Healthcare Center	Litchfield Investment Company, L.L.C.	Nexion Health at Thibodaux, Inc.	OHI Asset (LA), LLC	1300 LaFourche Drive Thibodaux, Louisiana 70301 504-446-1332 504-446-3974 (fax)	\$1,740,000
Vivian Healthcare Center	Litchfield Investment Company, L.L.C.	Nexion Health at Vivian, Inc.	OHI Asset (LA), LLC	912 S. Pecan Street Vivian, Louisiana 71082 318-375-2203 318-375-2866 (fax)	\$2,400,000

Pierremont Healthcare Center	Litchfield Investment Company, L.L.C.	Nexion Health at Pierremont, Inc.	OHI Asset (LA), LLC	725 Mitchell Lane Shreveport, Louisiana 71106 318-868-2789 318-868-6375 (fax)	\$5,880,000
Plainview Healthcare Center	West 24 th Street Associates, LLC	Nexion Health at Plainview, Inc.	OHI Asset (LA), LLC	2510 West 24th Street Plainview, Texas 79072- 1884 806-296-5584 806-296-9757 (fax)	\$5,293,000
lowa Park Healthcare Center	North Third Street Associates, LLC	Nexion Health at Iowa Park, Inc.	OHI Asset (LA), LLC	1109 North 3rd Street lowa Park, Texas 76367 817-592-4139 817-592-4799 (fax)	\$3,739,000
Midwestern Healthcare Center	Midwestern Parkway Associates, LLC	Nexion Health at Wichita Falls, Inc.	OHI Asset (LA), LLC	601 Midwestern Parkway Wichita Falls, Texas 76302- 2499 817-723-0885 817-723-0634 (fax)	\$5,828,000
Terrell Manor	North Francis Street Associates, LLC	Nexion Health at Terrell Manor, Inc.	OHI Asset (LA), LLC	1800 North Frances Street Terrell, Texas 75160 972-563-2652 972-563-0828 (fax)	\$4,565,000
Terrell Healthcare Center	West Nash Street Associates, LLC	Nexion Health at Terrell, Inc.	OHI Asset (LA), LLC	204 West Nash Street Terrell, Texas 75160-2607 972-563-7668 972-563-2769 (fax)	\$6,265,000

EXHIBIT D

LEGAL DESCRIPTIONS OF LAND

SCHEDULE 6.1.2

POWER AND AUTHORITY

NO CONFLICT

REAL ESTATE

LITIGATION

PERMITTED EXCEPTIONS

Zoning, use, and building laws, codes, regulations, and ordinances, and all other similar legal requirements, applicable to the Properties.

The state of facts the final updated Surveys and personal inspections of the Properties would show and are not objected to by Buyers under Section 7.4 hereof.

Real estate taxes, water charges, and sewer rents that become due and payable after the date of the Closing (which shall be subject to the apportionment provided for in this Agreement).

Unpaid installments of assessments that become due and payable after the date of the Closing.

All covenants, conditions, restrictions, and easements of record applicable to the Properties and not objected to by Buyers under Section 7.4 hereof.

Rights of utility companies to lay, maintain, install, and repair pipes, poles, conduits, cable boxes and similar and related utility facilities and equipment on, over, and under the Properties.

Rights, if any, of third-parties with respect to any portion of the Properties lying within the boundaries of a public or private road (subject to amendment or deletion upon approval of the final updated Surveys).

All matters described in the Commitments and the updated Surveys that are not objected to by Buyers in accordance with Section 7.4 hereof or are waived or are deemed waived by Buyers as a result of the purchase of the Properties at the Closing.

ENVIRONMENTAL MATTERS

RULE 13a-14(a)/15d-14(a) CERTIFICATION OF CHIEF EXECUTIVE OFFICER

- I, C. Taylor Pickett, Chief Executive Officer, certify that:
 - 1. I have reviewed this Quarterly Report on Form 10-Q of Omega Healthcare Investors, Inc.;
 - 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
 - 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
 - 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
 - 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 4, 2006

RULE 13a-14(a)/15d-14(a) CERTIFICATION OF CHIEF FINANCIAL OFFICER

- I, Robert O. Stephenson, Chief Financial Officer, certify that:
 - 1. I have reviewed this Quarterly Report on Form 10-Q of Omega Healthcare Investors, Inc.;
 - 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
 - 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
 - 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
 - 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 4, 2006

EXHIBIT 32.1

SECTION 1350 CERTIFICATION OF THE CHIEF EXECUTIVE OFFICER

- I, C. Taylor Pickett, Chief Executive Officer of Omega Healthcare Investors, Inc. (the "Company"), hereby certify, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350, that:
 - (1) the Quarterly Report on Form 10-Q of the Company for the three months ended June 30, 2006 (the "Report") fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
 - (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 4, 2006

/S/ C. TAYLOR PICKETT

C. Taylor Pickett Chief Executive Officer

EXHIBIT 32.2

SECTION 1350 CERTIFICATION OF THE CHIEF FINANCIAL OFFICER

- I, Robert O. Stephenson, Chief Financial Officer of Omega Healthcare Investors, Inc. (the "Company"), hereby certify, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350, that:
 - (1) the Quarterly Report on Form 10-Q of the Company for the three months ended June 30, 2006 (the "Report") fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended, and
 - (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 4, 2006

/S/ ROBERT O. STEPHENSON

Robert O. Stephenson Chief Financial Officer